# THE FISCAL THEORY OF THE PRICE LEVEL: A NARROW THEORY FOR NON-FIAT MONEY

Óscar J. Arce

Documentos de Trabajo N.º 0501

BANCODE**ESPAÑA** 

2005



THE FISCAL THEORY OF THE PRICE LEVEL: A NARROW THEORY FOR NON-FIAT MONEY

# THE FISCAL THEORY OF THE PRICE LEVEL: A NARROW THEORY FOR NON-FIAT MONEY

TOR NON-TIAT MONET
Óscar J. Arce (*)
BANCO DE ESPAÑA
(*) I wish to thank Marco Bassetto, José M. Martínez-Sánchez, Javier Vallés, an anonymous referee and, specially, my
supervisor, Nobuhiro Kiyotaki, and participants at several seminars at the LSE, Banco de España, IX Workshop on Dynamic Macroeconomics (Vigo) and 19th Congress of the EFA (Madrid) for helpful comments and discussions. All
errors are mine. The opinions and analyses herein are the responsibility of the author and, therefore, do not necessarily coincide with those of the Banco de España or the Eurosystem. Author's address: Research Department, Banco de España. Calle de Alcalá, 48. 28014 Madrid (Spain). Phone: +34 91 3386061. Fax: +34 91 3385678. e-mail: o.arce@bde.es
España. Calle de Alcalá, 48. 28014 Madrid (Spain). Phone: +34 91 3386061. Fax: +34 91 3385678. e-mail: o.arce@bde.es

The Working Paper Series seeks to disseminate original research in economics and finance. All papers have been anonymously refereed. By publishing these papers, the Banco de España aims to contribute to economic analysis and, in particular, to knowledge of the Spanish economy and its international environment.

The opinions and analyses in the Working Paper Series are the responsibility of the authors and, therefore, do not necessarily coincide with those of the Banco de España or the Eurosystem.

The Banco de España disseminates its main reports and most of its publications via the INTERNET at the following website: http://www.bde.es.

Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged.

© BANCO DE ESPAÑA, Madrid, 2005

ISSN: 0213-2710 (print) ISSN: 1579-8666 (on line)

Depósito legal:

Imprenta del Banco de España

#### **Abstract**

I examine the postulates of the Fiscal Theory of the Price Level (FTPL) under a nominal interest rate peg. First, I show that the usual definition of a non-Ricardian plan involves a number of government's non-credible policy commitments, thus confuting the interpretation of the FTPL as a policy-based equilibrium selection device. The main novelty of this criticism is that it is based on the same core assumptions maintained by this theory: there is a positive stock of governmentissued assets at the beginning of the history owned by the households, flow of funds constraints must be respected in every contingency, although transversality conditions may be violated at off-equilibrium prices.

Then I investigate some additional necessary conditions that allow the government to implement non-Ricardian fiscal plans that result in a unique equilibrium under an interest rate peg. A critical necessary condition for the credibility of such a fiscalist plan is that the equilibrium level of seigniorage must be non-positive. I argue that the fiscalist stock-analogy, under this monetary rule, is only meaningful, precisely, when money enters into the government constraint as a destination of funds, rather than as a source.

**Keywords:** Fiscal-monetary interactions, Fiscal Theory of the Price Level, Interest rate pegging, government commitment.

JEL classification: E31, E42, E61.

#### 1 Introduction

Sargent and Wallace (1981), in a pathbreaking article, showed a natural way in which fiscal and monetary policies are related to each other by highlighting the role played by the government's intertemporal budget constraint. One of the most important messages from the "Unpleasant Monetarist Arithmetic" is that, regardless of the policy regime at work, a certain degree of coordination between fiscal and monetary decisions is always needed, provided the government is willing to honor its policy announcements. Another critical observation drawn from Sargent and Wallace's story is that fiscal variables, such as taxes or public debt, can only affect nominal variables, such as prices, money supply or nominal interest rates, as long as the central bank accommodates its policy to satisfy some fiscal requirements, printing as much money (i.e. collecting seigniorage) as needed to cover a portion of the government's outlays, i.e., when the fiscal-monetary plan is conducted under a fiscal-dominance regime. Overall, in their economy the monetarist dictum goes on: the price level is always a monetary phenomenon, in spite of the eventual fiscal roots of the observed monetary stance.

Over the last decade, a number of economists<sup>1</sup> have challenged the above arguments concerning both the necessity of some degree of monetary and fiscal coordination and the monetary nature of the price level, developing the so-called Fiscal Theory of the Price Level (FTPL, henceforth). The cornerstone of this novel theory is the assumption that the government can commit to implement non-Ricardian policies (to be defined later). Taking this assumption as valid forces an important reconsideration of the main results of Sargent and Wallace's. First, under a non-Ricardian policy, the policy coordination problem vanishes: fiscal and monetary policies can be designed in a completely uncoordinated fashion without it meaning that the government violates any budget constraint in equilibrium. Second, the previous monetarist dictum breaks down: under the fiscalist postulates the price level and inflation are, fundamentally, fiscal phenomena, with money playing a secondary role.

The fact that under a non-Ricardian policy the price level is directly influenced by fiscal variables (e.g. the stock of government debt and the sequence of primary surpluses/deficits) can be exploited to design policies which are supposed to remove the classic nominal indeterminacy problem associated with a pure nominal interest rate peg and the multiplicity of equilibria under an exogenous money supply rule. Over the last few years, a growing number of papers have proliferated in which the intellectual framework of the FTPL is employed to provide answers to some practical macroeconomic questions such as liquidity traps, hyperinflations, currency crisis, international monetary policy coordination problems, questions related to the fiscal policy design in monetary unions, the effectiveness of the independent central bank paradigm, etc.<sup>2</sup>.

<sup>&</sup>lt;sup>1</sup>Some of the seminal works on the FTPL include the following: Leeper (1991), Sims (1994) and Woodford (1995). Other contributions supporting its theoretical foundations are contained in Bassetto (2002), Cochrane (1998, 2003a, 2003b), Daniel (2003, 2004), Davig *et al.* (2004), Gordon and Leeper (2002), Sims (1999, 2002) and Woodford (1998, 2001). Christiano and Fitzgerald (2000), and Kocherlakota and Phelan (1999) contain clear explanations of the basic FTPL's arguments.

<sup>&</sup>lt;sup>2</sup>Benhabib et al. (2002) study alternative policies, including non-Ricardian ones, to avoid liquidity

Given the long-reaching implications of the FTPL, the validity of its basic postulates has attracted much attention. Among the advocates of the FTPL, it is usually argued that the government is not limited in its actions by any intertemporal budget constraint, that the real value of fiat government-issued money can be determined in a fiscalist world as the price of a private firm's equity (this is the so-called stock-analogy, defended by Cochrane (2003a, 2003b) and Sims (1999)) and, in general, that a non-Ricardian policy is theoretically as valid as it is a Ricardian one, with the assumptions underlying the former being neither unfeasible nor unrealistic (e.g. Daniel (2003, 2004), Woodford (1998, 2001)). Some critiques of the FTPL point towards the implausibility of non-Ricardian policies, since these policies imply that the government is allowed to violate its intertemporal budget constraint at off-equilibrium prices (Buiter (1999, 2002)), the incompatibility of non-Ricardian policies with the requirements imposed by the Walrasian competitive equilibrium concept (Marimón (2001)), the possibility of government default under a non-Ricardian fiscal policy (Cushing (1999)) or the unrealistic assumption on the existence of a non-zero stock of nominal government-issued assets at the "beginning of the world" (Niepelt (2004)).

Bassetto (2002) suggests that much of the current confusion about the FTPL is mainly due to two particular reasons. First, he argues that the Walrasian framework employed in all the previous papers in this area is not the appropriate one to deal with a theory which critically depends on some assumptions about the off-equilibrium dynamics of the economy. Second, he makes the point that the usual way in which non-Ricardian policies are defined is not a "correct" one. Bassetto frames a simple non-monetary model, exploiting the so-called cash-less economy assumption, into a non-Walrasian economy. His main results are summarized in the following passage: "I show that there exist government strategies that lead to a version of the fiscal theory, in which the price level is determined by fiscal variables alone. However, these strategies are more complex than the simple budgetary rules usually associated with the fiscal theory [...]".

In this paper, I identify a minimum set of conditions that guarantee the internal consistency of the FTPL's arguments, under the assumption that the government commits to maintain a nominal interest rate peg, supplying as much money as demanded by the private sector<sup>3</sup>, a policy that has received much attention in earlier works in this area. Some of the arguments developed here resemble one of the observations made by Bassetto: the design of a fiscal-monetary program in the spirit of the FTPL's non-Ricardian policies is not as straightforward as it usually assumed. However, this paper departs form Bassetto's one in two main respects. First, I allow individuals to hold money, by introducing real balances as an additional argument in the utility function and, second, I cast my arguments in a standard Walrasian framework widely employed by an

traps. Loyo (2001) describes some hyperinflationary episodes in Brazil using the postulates of the FTPL. Daniel (2001b), Mackowiak (2003) and Corsetti and Mackowiak (2003) explain currency crisis along the fiscalist arguments. Andrés et al. (2002), Daniel (2001a), Dupor (2000) and Sims (1997) explore the implications of the FTPL in an open economy. Bergin (2000) and McCallum (1999) discuss the fiscal-monetary coordination problem in a monetary union in light of the FTPL's arguments. Afonso (2002), Canzoneri et al. (2001) and Sala (2002) discuss the empirical relevance of the FTPL.

<sup>&</sup>lt;sup>3</sup>In a related paper, Arce(2004), I study the validity of the FTPL when the central bank directly fixes the nominal supply of money.

important number of previous authors (both proponents and opponents of the FTPL). The first departure is rooted on an obvious observation: in an economy in which money does not exist, one cannot get any meaningful result on how the real value of money is determined. The second departure is of technical nature. Firstly, I argue that the analysis of the fundamental "weak points" of the FTPL can be carried out within a Walrasian framework, for some critical questions in this analysis, such as the fulfilment of flow of funds constraints and individual rationality, are not specific of the equilibrium concept employed or of the particular assumptions about how prices are formed. Secondly, I show that, under some conditions, the government can credibly commit to maintain some exogenously targeted fiscal-monetary policy which is consistent with a unique equilibrium price sequence, in the spirit of the FTPL, by implementing a price-contingent fiscal plan in a standard Walrasian economy.

I first argue that the validity of the FTPL's argument needs a mechanism relating the individual's optimal decisions to the price level and that the standard definition of a non-Ricardian policy assumes that such a mechanism exists without proving it. Moreover, I show that, according to the most basic arguments of the standard FTPL, such a theory can never perform as a policy-based equilibrium selection device in the context of the economy and the policy rule studied here. In doing so, I add a number of new arguments to the existing critiques against the FTPL. Then I define a minimum set of conditions that allow the government to credibly commit to a fiscal-monetary plan which results in a unique equilibrium by creating a link between prices, policies and individual optimal behavior. A critical necessary condition for a credible government policy, that includes a monetary plan not involving a direct control of a monetary aggregate, to yield a unique equilibrium price sequence is that the equilibrium level of seigniorage cannot be positive, i.e., except in the particular case in which seigniorage is equal to zero, the government must redeem a fraction of the existing stock of money in exchange for consumption goods. This observation implies that the fiscalist arguments under a nominal interest rate peg, as they are usually presented in the literature, are not generally consistent with the notion of fiat (i.e. non-convertible) money, or, equivalently, a properly defined non-Ricardian plan requires dollars (rather than bonds) to be net wealth. This is, perhaps, the main result of this paper.

Some other questions approached here include the following. I show that, contrary to the standard FTPL's fiscal-monetary independence proposition, a credible fiscalist policy must always involve a certain degree of coordination. I also provide a critical revision of the general validity of the comparison between the FTPL's arguments and the standard financial asset-pricing theory. In particular, I argue that there is no analogy between the standard FTPL and the asset-pricing theory, although there is a similarity between the mechanism underlying the determination of the real value of a firm's stock and the one underlying the determination of the equilibrium price sequence under a credible (i.e. with convertible money) non-Ricardian fiscal-monetary plan.

The paper is structured as follows. Section 2 describes the general set-up of the monetary economy studied here and presents a basic "canonical" example of a non-Ricardian fiscal-monetary program. Section 3 explores in detail the problems associated with the standard definition of a non-Ricardian policy using a simple one-period model.

Section 4 describes a set-up in which the government can credibly commit to implement non-Ricardian policies. Section 5 deals with the fiscalist stock-analogy. Section 6 contains the extension of the main results to a multi-period economy. Section 7 summarizes the main conclusions of the paper.

## 2 The canonical non-Ricardian argument

In this paper I consider a simple representative agent model with exogenous and constant endowments of consumption goods. Time is discrete and there is no uncertainty. Money is introduced as an argument in the household's utility function. Thus, in its basic aspects, the economy studied here is similar to the ones considered by some advocates (e.g. Daniel (2003, 2004), Leeper (1991) and Woodford (1995, 1998, 2001)), and opponents (e.g. Buiter (2001, 2002), Cushing (1999), Niepelt (2002) and McCallum (2001)) of the FTPL.

#### 2.1 The model

#### 2.1.1 The households

The households form their expectations rationally and try to maximize the following objective function<sup>4</sup>

$$\max_{\{c,M\}} \sum_{t=1}^{\infty} \beta^{t-1} \left[ u(c_t) + v(M_t/P_t) \right]$$
 (1)

where u and v are increasing and concave functions in their respective arguments and satisfy the Inada conditions. The function u captures the utility enjoyed by the household from consuming a certain amount of goods,  $c_t$ , and v stands for the liquidity services provided by real money holdings, i.e. nominal balances,  $M_t$ , deflated by the general price level,  $P_t$ . The parameter  $\beta$  is the subjective discount factor and satisfies  $0 < \beta < 1$ .

The representative household's flow budget constraints are

$$\frac{B_t + M_t}{P_t} \le y - \tau_t - c_t + \frac{(1 + i_{t-1}) B_{t-1} + M_{t-1}}{P_t} \quad (t \ge 1)$$
 (2)

where y is the (positive) exogenous household's endowment of consumption goods,  $\tau_t$  is a lump-sum tax payable to the government;  $B_t$  is the amount of dollar-denominated government bonds which are assumed to mature one period after issued. Each bond issued at t-1 is sold at a price of one dollar and redeemed at a value equivalent to  $1+i_{t-1}$  dollars at t. The household may also be endowed with an initial stock of financial wealth: some dollar-denominated government bonds, inclusive of interest,  $(1+i_0)B_0$ , and some money balances,  $M_0$ . The value of each of these two terms is given exogenously and assumed to be non-negative. Also, the households are precluded from issuing money, i.e.  $M_t \geq 0$ . For the ease of the exposition, it is assumed that the households do not issue debt either<sup>5</sup>, i.e.  $B_t \geq 0$ . In choosing a path for lifetime consumption and money holdings, the household is restricted by the following transversality condition

<sup>&</sup>lt;sup>4</sup>In some of the following sections I use some restricted versions of the infinite horizon model described in this section (e.g. one-period and two-periods economies). Those restricted versions can be readily derived as special cases of this more general framework.

<sup>&</sup>lt;sup>5</sup>This assumption does not affect the generality of the arguments in this paper. See Bassetto (2002) for a similar constraint.

$$\lim_{T \to \infty} \left( M_{T-1} + (1 + i_{T-1}) B_{T-1} \right) / \prod_{s=1}^{T-1} (1 + i_s) \ge 0$$
 (3)

I define the household's utility maximization problem as his choice of a lifetime consumption and portfolio paths<sup>6</sup>  $\{c_t, M_t, B_t\}$  satisfying (2), the non-negativity constraints on assets holdings, given an initial stock of financial wealth, the price and interest rate sequences  $\{P_t, i_t\}$ , and the income and tax sequences  $\{y, \tau_t\}$ . The following conditions are necessary and sufficient for an optimal choice of consumption, money and bonds holdings plans.

When the household behaves optimally, i.e. not leaving any intrinsically valuable resources unconsumed, both the set of budget constraints (2) and the limiting condition (3) hold as equalities, in which case we learn that the following present-value constraint necessarily holds

$$(1+i_0) B_0 + \sum_{t=1}^{\infty} \frac{P_t (y-\tau_t)}{\prod_{s=1}^{t-1} (1+i_s)} = \sum_{t=1}^{\infty} \frac{P_t c_t + M_t - M_{t-1}}{\prod_{s=1}^{t-1} (1+i_s)}$$
(4)

This optimality condition states that the total value of the household's wealth at any period  $t \ge 1$ , inclusive of maturing government bonds plus current and future endowments net of taxes be equal to the total value of the household's expenditure which includes consumption and seigniorage (i.e. purchases of government-issued money).

The following standard first order conditions complete the characterization of the solution for this maximization problem

$$\frac{v'(m_t) - u'(c_t)}{P_t} + \beta \frac{u'(c_{t+1})}{P_{t+1}} = 0$$
 (5)

$$\beta (1 + i_t) \frac{u'(c_{t+1})}{P_{t+1}} - \frac{u'(c_t)}{P_t} = 0$$
 (6)

where  $m_t \equiv \frac{M_t}{P_t}$  is the household's demand for real balances. Unless otherwise stated, it is also assumed that the felicity function v satisfies the following inequality

$$\lim_{m \to 0} mv'(m) > 0 \tag{7}$$

In view of the above inequality, we learn that speculative hyperinflations are not possible equilibrium outcomes<sup>7</sup>.

<sup>&</sup>lt;sup>6</sup>Notice that there is another choice faced by the household at every period: whether to redeem his entire stock of maturing bonds in exchange for the goods and assets (money and/or newly issued bonds) given by the government. Since maturing bonds are assumed to expire after their maturity date, it will be assumed throughout that the household always redeems his entire stock of initial bonds.

<sup>&</sup>lt;sup>7</sup>See Obstfeld and Rogoff (1983). This assumption is made here for simplicity as the main arguments in this paper do not depend on the possibility of rational speculative paths. In Arce (2004) I analyze some speculative solutions in the context of the FTPL.

#### 2.1.2 The government

The government in this economy sets the level of taxes and public consumption and manages the public debt, issuing and redeeming bonds, fixing the nominal interest rate and supplying money endogenously in the amount required by the private sector so as to accommodate the "needs of trade". The consolidated government's sequence of flow of funds constraints is given by

$$\frac{B_t^s + M_t^s}{P_t} = g_t - \tau_t + \frac{M_{t-1}^s + (1 + i_{t-1}) B_{t-1}^s}{P_t} \qquad (t \ge 1)$$
 (8)

where  $B_t^s$  and  $M_t^s$  are, respectively, the government supply of bonds and money and  $g_t$  stands for government consumption and satisfies  $g_t < y$ . Iterating forward the constraint (8) and imposing the following transversality condition

$$\lim_{T \to \infty} \left( M_{T-1}^s + (1 + i_{T-1}) B_{T-1}^s \right) / \prod_{s=1}^{T-1} (1 + i_s) = 0$$
 (9)

gives the government intertemporal budget constraint

$$(1+i_0) B_0 + \sum_{t=1}^{\infty} \frac{P_t g_t}{\prod_{s=1}^{t-1} (1+i_s)} = \sum_{t=1}^{\infty} \frac{P_t \tau_t + M_t^s - M_{t-1}^s}{\prod_{s=1}^{t-1} (1+i_s)}$$
(10)

which holds if and only if (8) and (9) are simultaneously satisfied.

### 2.2 Ricardian and Non-Ricardian policies

Bellow I introduce a standard definition of Ricardian and non-Ricardian policies and examine the implications of alternative fiscal-monetary programs for the equilibrium of the model, using the paradigm of the Walrasian competitive equilibrium. The aim here is to make a description of the implications of these two alternative classes of policies in terms of the (in)determinacy of the equilibrium price sequence, as they are usually presented in the previous literature. I use the term "canonical" here to refer to the class of solutions of the model in which it is explicitly assumed that the government always honors its policy-announcements, (i.e. redeeming its eventual outstanding obligations at their contractual value and meeting its targeted sequences of taxes and government consumption). Whether this a sensible assumption to be held under any circumstance or not is at the heart of the current discussion on the validity of the FTPL's postulates, an issue which is analyzed in detail in the following sections.

**Definition 1 (Canonical non-Ricardian vs Ricardian distinction)** A policy is Ricardian if it is formulated in such a way that the government intertemporal constraint (10) is satisfied for any price sequence  $\{P_t\}_{t=1}^{\infty}$ . It is non-Ricardian if it is only satisfied in equilibrium. Equivalently, a policy is Ricardian if the transversality condition (9) is satisfied for any price sequence and non-Ricardian if it is only satisfied in equilibrium<sup>8</sup>.

<sup>&</sup>lt;sup>8</sup>This definition corresponds to the one given by Woodford (1995). Although Woodford explicitly refers to equation (10), the fact that the flow of funds constraint (8) holds as an identity allows for an equivalent definition for a non-Ricardian policy based on the transversality condition (9).

In examining the cases presented below, I focus on the following particular, although widely used in the literature, Ricardian and non-Ricardian policies<sup>9</sup>:

- 1. Ricardian interest rate peg (RP): the government sets a price-invariant level of public consumption for each period,  $\{g_t\}$ , and commits to a sequence of non-negative interest rates  $\{i_t\}$ , which is also assumed to be independent of the observed price-level.
- 2. Non-Ricardian interest rate peg (NRP): the government chooses the same policy-instruments and money supply rule as in RP and also commits to a price-invariant sequence of taxes,  $\{\tau_t\}$ , without any feedback from the observed price level.

The definition of a competitive equilibrium for this economy is the following:

**Definition 2** A perfect foresight competitive equilibrium in this economy is a set of allocations  $\{c_t\}$ ,  $\{g_t\}$ ,  $\{B_t\}$ , and  $\{M_t\}$ , price and interest rate sequences  $\{P_t, i_t\}$ , and a sequence of taxes  $\{\tau_t\}$  such that the following conditions are satisfied:

- 1. The government satisfies its budget constraint (8) as an equality in every single period and the transversality condition (9).
- 2. Households maximize their utility subject to subject to the constraints (2) and (3) and the non-negativity constraints  $M_t \geq 0$  and  $B_t \geq 0$ , given the sequences  $\{P_t, i_t\}$  and the government policy.
- 3. All markets clear in every period, i.e.,  $y = c_t + g_t$ ,  $M_t = M_t^s$ , and  $B_t = B_t^s$ .

#### 2.2.1 The Ricardian canonical solution

Here I study how equilibrium prices are determined under the hypothesis that the government intertemporal budget constraint (10) is always binding, regardless of the value taken by the endogenous variables of the model. The following proposition contains the main results with respect to the (in)determinacy of the equilibrium price sequence.

**Proposition 1** Under the Ricardian policy RP, the are a continuum of equilibrium price sequences.

**Proof.** Given the set of government's commitments, a competitive equilibrium is characterized by the following conditions:

- (i) The first order conditions (5) and (6).
- (ii) The household's flow of funds constraint (2) and the transversality condition (3) hold as equalities.

<sup>&</sup>lt;sup>9</sup>The main arguments of this paper do not hinge on the particular design of Ricardian and non-Ricardian policies, as it will become clear later. Focusing on the simple policy rules described here, however, adds to the clarity of the exposition. All the sequences are understood to be defined for  $t \ge 1$ .

- (iii) The government implements price-invariant sequences of public consumption and nominal interest rates, according to the definition of the policy RP given above, adjusting the sequence of taxes so as to satisfy (10) and (8) for any price sequence.
  - (iv) All the markets (consumption goods, money and bonds) clear.

Let's consider an arbitrary positive and finite initial price level  $P_1^*$ . It can be verified that there exists a competitive equilibrium associated with that initial price level. As the government sets the sequence of public consumption without any feedback from the price level, we learn that in equilibrium private consumption satisfies  $c_t = y - g_t$  for all  $t \ge 1$ . Then, the government's commitment to an exogenous sequence of nominal rates implies that, according to the first order conditions (5) and (6), the optimal household's demand for nominal balances in the initial period t = 1, in equilibrium, satisfies

$$M_1(P_1^*) = v'^{-1} \left[ \frac{i_1}{1 + i_1} u'(y - g_1) \right] P_1^*$$
 (11)

Given that particular  $P_1^*$ , the government sets a sequence of taxes satisfying (10), i.e. the whole tax-sequence satisfies

$$\frac{(1+i_0)B_0}{P_1^*} + \sum_{t=1}^{\infty} \frac{g_t}{\prod_{s=1}^{t-1} (1+r_s)} = \sum_{t=1}^{\infty} \frac{\tau_t + \frac{M_t^s - M_{t-1}^s}{P_t}}{\prod_{s=1}^{t-1} (1+r_s)}$$
(12)

where  $r_s = (1 + i_s) (p_s/p_{s+1}) - 1$  is the real rate of return on bonds. Then, for a particular first element of the tax sequence, call it  $\tau_1(P_1^*)$ , there is a unique end-of-period stock of government bonds satisfying the budget constraints (2) and (8) and the market clearing conditions at t = 1, given by

$$B_1(P_1^*) = P_1^*(q_1 - \tau_1(P_1^*)) - M_1(P_1^*) + M_0 + (1 + i_0)B_0$$

The equilibrium price level for the second period can then be solved uniquely from (6), and applying the same steps above, the entire collection of the equilibrium sequences for the endogenous variables (i.e. private consumption, stock of nominal money, stock of government bonds, taxes and prices) can be solved recursively for an arbitrary  $P_1^*$ .

The main message from this Ricardian fiscal-monetary program is that, provided we assume that the present-value constraint (10) must hold regardless of the particular values of the endogenous variables, the government must react to the real value of the initial stock of debt by adjusting the sequence of taxes as required (by (12)), so as to avoid default or supersolvency premium on the initial stock of debt. In other words, according to the SW's "game of the chicken" parable, as the fully elastic money supply rule at work leaves the equilibrium price sequence undetermined, fiscal policy must "blink".

#### 2.2.2 The non-Ricardian canonical solution

The following proposition contains the main implications arising from the assumption that the government implements a plan without any degree of fiscal-monetary coordination, as indicated by the non-Ricardian program defined above.

**Proposition 2** Under the non-Ricardian policy NRP, the equilibrium price sequence is unique provided a sign condition holds.

**Proof.** Combining the market clearing and the household's optimality conditions, including the transversality condition (3) holding as an equality, we learn that in equilibrium the following equation must hold in the initial period

$$\frac{M_0 + (1 + i_0) B_0}{P_1} = \sum_{t=1}^{\infty} \frac{\tau_t - g_t + \frac{i_t}{1 + i_t} m_t}{\prod_{s=1}^{t-1} (1 + r_s)}$$
(13)

Notice that every term in the RHS of (13) is independent of the price level,  $P_1$ , under this policy. Some elements are directly fixed by the government ( $\tau_t$  and  $g_t$ ),  $r_s$  can be derived by combining the exogenously set fiscal instruments together with the market clearing and individual optimization conditions, while  $M_0$  and  $(1 + i_0) B_0$  in the LHS are given at the beginning of period 1. It follows that under this set of policy-choices the price level in the initial period,  $P_1$ , is uniquely determined by equation (13). For this price to be positive and finite, the following sign condition must hold

$$0 < \sum_{t=1}^{\infty} \frac{\tau_t - g_t + \frac{i_t}{1+i_t} m_t}{\prod_{s=1}^{t-1} (1+r_s)} < \infty$$
 (14)

The above sign-condition is the only restriction placed upon the sequence of taxes and government consumption<sup>10</sup>. Once the price level for the initial period is pinned down, the entire set of sequences of endogenous variables is uniquely determined too. The equilibrium money supply in the initial period,  $M_1^S$ , can be obtained from (5), then  $P_2$  can be solved uniquely from (6), with the supply of government debt in real terms,  $\frac{B_1^S}{P_1}$ , obeying the sequence of budget constraints (8) and so on.

Clearly, assuming that the economic authorities can commit to a non-Ricardian policy like the one outlined in the previous proposition results in a considerable "gain" in terms of the (lower) number of potential equilibria. Treating the intertemporal constraint (10) (or the transversality condition (9)) as an independent equilibrium condition rather than as an identity, which must hold always, provides an "extra" equation that can be exploited through the implementation of a non-Ricardian policy so as to solve the "classical" nominal indeterminacy problem described in Proposition 1. The main practical implications, from the perspective of the policy-mix design, arising from Proposition 2 is that fiscal-monetary coordination is not longer needed: if neither player is blinking the price level will adjust.

In providing supportive arguments for the fiscalist (unique) equilibrium constructed before, the proponents of the FTPL justify the interpretation of (13) as an additional separate equilibrium condition either by accepting that the government is limited on its actions by an intertemporal budget constraint, but with the extra qualification that the government is a "big player", so that it need not to take prices as given (see Woodford (1998, 2001)), or even by denying that such a constraint exists at all, arguing that (13)

<sup>&</sup>lt;sup>10</sup>For the remaining of the paper, I will assume that this sign condition holds unless otherwise stated.

must be interpreted as a "government valuation equation" rather than as a constraint on its actions. Under this approach, the government can be viewed as a private firm whose equity is priced according to the future stream of profits, where an analogy is drawn between a firm's equity and the nominal-denominated government's obligations (the term  $(1 + i_0) B_0 + M_0$ ), and between the firm's profit stream and the government's total surpluses (the RHS of (13)). From this perspective, the plausibility of non-Ricardian policies is accepted as it is the general idea that the future profits of a firm need not to be influenced by the current price of its shares and, therefore, the transversality condition is only satisfied when the stock of the firm is "correctly" valued (i.e. at equilibrium prices). Further, if money and nominal debt can be understood as the firm's stock and, hence, as residual claims to government total surpluses, the issue of default on government-issued assets is irrelevant, simply because it makes no sense to talk about default on shares. Thus, the non-default assumption embedded in the "canonical" case presented here would be plainly justified. This stock-analogy argument can be found, e.g., in Cochrane (2003a, 2003b), Christiano and Fiztgerald (2000) and Sims (1999).

# 3 Transversality conditions: Assumptions vs theory

This section explores the internal consistency of the assumptions underlying the non-Ricardian canonical case discussed in Proposition 2. Some other authors have shown their concern about the validity of the FTPL under an interest rate peg, as well. Buiter argues that, as the government commits to run a completely exogenous fiscal policy, there is no reason why the sign condition (14) will hold with generality: "The most the fiscal theory of the price level therefore could aspire to, when the arbitrary (sign condition) is satisfied, is to be a way of removing the price level indeterminacy characteristic of equilibria under a Ricardian nominal interest rule, when nominal prices are flexible" (Buiter (1999), parentheses added). Niepelt (2004) shows his concern about the assumption of a non-zero initial stock of nominal assets: "In this paper, I offer a resolution to this debate. The fundamental problem of the FTPL is that the feasibility of non-Ricardian policy hinges on the assumption of non-zero initial nominal government liabilities. This assumption is not well founded [...]". Cushing (1999) argues that, in face of an initial stock of government debt, the possibility of default in the first period breaks down the fiscalist uniqueness result: "In this section, I point out that admitting the possibility of fiscal default shows the price level to be indeterminate under an interest rate rule".

Notwithstanding the relevance of the above criticisms, the arguments given in this section point towards a different direction. Kocherlakota and Phelan (1999) argue that the FTPL, at its core, contains prescriptions about the government behavior at offequilibrium prices, an issue which is not testable, thus rendering the question on the validity of the FTPL as an equilibrium selection device "a religious, not a scientific issue". Below, I argue that it is indeed possible to analyze the arguments given by the FTPL regarding the behavior of the economic agents in those situations that this theory claims to be off-equilibrium outcomes, even after recognizing that this question is not testable. In doing so, I exploit a feature of a critical element in the model: flow of funds constraints must always hold (in- and off-equilibrium) regardless of the identity, size and behavior of the corresponding agent. Such a status for the flow of funds constraints, to the best of my knowledge, has not been challenged by any proponent of the FTPL<sup>11</sup>. Interestingly, the analysis pursued here does not need to give such a status to the transversality conditions and conforms the fiscalist view that those conditions are, indeed, only equilibrium conditions. The role of the transversality conditions have been placed at the center of the debate. Treating them as "equilibrium" conditions is at the core of non-Ricardian policies, as stated in Definition 2. For some critics of this theory (e.g. Buiter), this is not an admissible step. By keeping on the interpretation given by the proponents of the FTPL, I aim to provide a clearer and less controversial exposition of the weak points of this theory.

In particular, I show that the equilibrium uniqueness result in Proposition 2 is not an outcome driven by government's policies, but rather by some special assumptions about the households' decisions. In this sense, it can be argued that the FTPL is not a policy-based equilibrium selection device. And this proposition goes on even when the

<sup>&</sup>lt;sup>11</sup>An explicit treatment of the government flow of funds constraint (8) as an *identity* can be found, for instance, in Bassetto (2002), Cochrane (2003b), Davig *et al.* (2004) and Sims (2002).

economist does not observe the off-equilibrium implementation of the particular equilibrium selection device at work, something which is necessarily the case when in face of a successful device, like e.g. the one designed by Obstfeld and Rogoff (1983) to rule out speculative hyperinflations.

To keep the exposition as simple as possible, I first consider a one-period model<sup>12</sup>, in which the representative household holds an initial stock of financial wealth consisting on some dollars and bonds. As in the general model presented before, the household receives an endowment, y, and tries to maximize his utility by choosing how much to consume, how many dollars to hold at the end of the period and how many bonds, maturing after death, to buy. The government sets the level of taxes,  $\tau_1$  and public consumption,  $g_1$ . Notice, that within this narrow one-period framework there is no room for the government to commit to a non-zero effective redemption-value for bonds issued this period. However, for the arguments defended here, the critical fact concerning monetary policy is the commitment to elastically supply as much money as demanded, rather than the eventual announcement of a particular next period's redemption value payable by government's bonds<sup>13</sup>. The extension of the main results obtained within this one-period framework to a multi-period setting, where the government announces a particular interest rate is straightforward, as shown later.

The household is constrained in his choices by the following flow of funds and no-Ponzi games constraints

$$c_1 \le y - \tau_1 + \frac{M_0 + (1 + i_0) B_0 - M_1 - B_1}{P_1}$$
 (15)

$$\frac{B_1}{P_1} \geq 0 \tag{16}$$

The optimal behavior of the household is characterized by the following conditions:

$$u'(c_1) = v'\left(\frac{M_1}{P_1}\right) \tag{17}$$

$$\frac{B_1}{P_1} = 0 (18)$$

Equation (17) implies that the individual optimally chooses consumption and real balances equating their respective marginal utilities. Condition (18) follows from the "end of the world" assumption: as bonds issued in period 1 mature one period after, no rational agent would be willing to give up some valuable resources for that debt. An additional necessary condition for optimization is that the budget constraint (15) holds as an equality.

<sup>&</sup>lt;sup>12</sup>This simplification can also be found in Christiano and Fiztgerald (2000). Cochrane (2003b) also provides some examples in one-period economies.

<sup>&</sup>lt;sup>13</sup>At this point, it is worth noticing that what drives the multiplicity of equilibria result in the Ricardian case discussed in Proposition 1 is the mathematical property of homogeneity of degree zero in the equilibrium demand for real balances. The one-period framework preserves that feature and, not surprisingly, yields similar "canonical" results as in Propositions 1 and 2.

The consolidated government's budget constraint is

$$\frac{B_1^s + M_1^s}{P_1} = g_1 - \tau_1 + \frac{M_0 + (1 + i_0) B_0}{P_1}$$
 (19)

Regarding the monetary rule at work, it is assumed that the government commits to supply as many dollars as demanded at the ongoing price level.

According to Proposition 2, if the government commits to set  $\tau_1$  and  $g_1$  without any feedback from the observed price level, the following equation, which is derived by combining (17), (18), (19) and the market-clearing conditions, gives the unique fiscalist equilibrium price level,  $P_1^F$ ,

$$\frac{M_0 + (1 + i_0) B_0}{P_1^F} = \tau_1 - g_1 + m_1 \tag{20}$$

where  $m_1 = v'^{-1} [u'(y - g_1)]$ . The FTPL's logic for this uniqueness result is, at first sight, simple. Any price different from  $P_1^F$  above will result in the violation of an individual optimal condition or/and a market clearing condition, for these are, in principle, the only ingredients included in the characterization of the fiscalist equilibrium (see proof of Proposition 2). In what follows I check the general validity of this argument. In doing so, I follow a simple strategy: pick up an arbitrary price  $\overline{P}_1 \neq P_1^F$  and analyze how, if at all, such a price is inconsistent with the three equilibrium requirements listed in Definition 2: (i) fulfillment of government constraints, (ii) individual optimization problem and (iii) market-clearing.

To clear the desk, it is convenient to recall the following observation: when dealing with a failure of the market clearing conditions, we can concentrate just on situations in which the markets for goods and for new bonds fail to clear since, given the fully elastic monetary rule considered here, we learn that the supply of dollars is always identical to the demand.

Firstly, let's consider several alternative interpretations about the violation of some of the equilibrium requirements listed above when considering an arbitrary price  $\overline{P}_1 < P_1^F$ . For the sake of clarity, it is convenient to consider the two central pieces of the intuitive fiscalist argument given above, market clearing and individual optimization, separately.

(A market clearing approach). Let's assume that the households never demand a positive quantity of non-performing bonds, regardless of the price level, i.e.  $\frac{B_1}{P_1} = 0$   $\forall \overline{P}_1$ . Notice that such a lending rule is always optimal. Given that lending rule, we can solve for the household's goods and dollars demand functions combining the budget constraint (15), holding as an equality after imposing a zero-final stock of desired bonds, and the first order condition (17). Nothing up to this point presumes an underlying non-optimal behavior of the households. In face of  $\overline{P}_1 < P_1^F$ , we see from the government budget constraint (19) that the households refusal to purchase new bonds implies that the government cannot attain its fiscal objectives simultaneously, thus, unchaining a potential "crisis". A natural next step is to consider the possibility that the government activates a crisis-resolution device so as to render  $\overline{P}_1$  a non-equilibrium outcome, hence, defeating the crisis. There are not many available options for a government who has

already committed to some no price-contingent monetary and fiscal policies, apart from printing bonds and trying to sell them. This, assuming is a costless activity, is always feasible for the government. Then, one could think that a positive supply of government bonds, when coupled with a zero demand, creates a problem of excess of supply in the newly-issued bonds market or, alternatively, an excess of demand in the goods market<sup>14</sup>. Let's treat these two markets (bonds and goods) separately.

In the bonds market, the government, as a monopolist, may take two courses of action: it may fix the quantity of bonds,  $B_1^s$ , or it may announce the price at which it will meet the demand. This latter form of government intervention is, perhaps, the most popular in macro models, specially when assuming that the government chooses the price of a dollar as the reference-price at which it supplies its bonds<sup>15</sup>, say by setting the price of one bond supplied today equal to the price of one dollar (i.e.  $\overline{P}_1$ ; this is the convention followed in this paper). In this case there exists a unique market-clearing quantity consistent with a zero real demand for bonds,  $B_1 = B_1^s = 0$ . On the other hand, if the government chooses the quantity of bonds to be supplied at each bond-price, say  $\frac{1}{Q_1}$ , then there is a unique market clearing bond-price,  $\frac{1}{Q_1} = 0$ . Alternatively, we might think of the government offering contracts specifying a given quantity of bonds to be delivered at a particular dollar-price, of the kind  $(B_1^s, P_1^s)$ , letting the "market forces" determine the price of each contract. Still, the market for debt-contracts may clear at a zero real value for each contract offered. To put it in plain words, the market for a valueless asset (from the perspective of the potential buyers) may always "clear": either that asset is not traded at all or, if one wishes so, it is transacted at a zero real value.

Now, given that the household never wastes resources purchasing non-performing bonds when these are offered at a positive price, we learn that the government demand for consumption satisfies the following equality

$$g_1 = \tau_1 - \frac{M_0 + (1 + i_0) B_0 - M_1}{\overline{P}_1}$$
 (21)

Actually, in writing (21), we do not need to assume that the bonds market would "clear" according to the argument in the paragraphs above, since, the fact that the households never lend resources to the government, implies that the government cannot posit a demand for goods greater than its available resources at that price (i.e. the terms in the RHS of (21)). That is, for writing a well-posed demand function, we need to consider the amount of wealth that the agent can seize in supporting its bids. In view of (21), when the government commits to a fixed  $\tau_1$ , total wealth available for government consumption increases with the price level and vice versa, i.e. a commitment to a given  $g_1$  is only

<sup>&</sup>lt;sup>14</sup>This "disequilibrium" outcome becomes apparent by summing up the household's constraint (15), holding as a equality with  $\frac{B_1}{P_1} = 0$  and the government's constraint (19) after imposing  $\frac{B_1^s}{P_1} > 0$  and  $M_1 = M_1^s$ .

<sup>&</sup>lt;sup>15</sup>The fact that the government is only choosing the price of reference at which it is willing to supply bonds and not choosing the exact value of that price directly, does not mean that the government is not choosing the supply-price. Private banks usually fix the interest rate at which they are willing to meet the demand for loans by using a reference rate (e.g. LIBOR), which, of course, cannot be understood as the private banks as having their hands free to simultaneously choose interest rates and transacted quantities.

possible if  $\tau_1$  varies (inversely) with the price so as to satisfy (21) as an identity<sup>16</sup>. Once total wealth available for government consumption is computed correctly, i.e. accounting for the fact that the households will not purchase non-performing bonds, the ability of the government to preclude any  $\overline{P}_1 < P_1^F$  as an equilibrium outcome by printing bonds so as to induce an excess-of-demand in the goods market vanishes<sup>17</sup>.

In sum, a government's plan to fight a crisis based upon its unlimited ability to print non-performing bonds will not preclude the crisis whenever the households behave in a contingent optimal behavior. Further, when printing bonds is the  $unique^{18}$  alternative government plan, a crisis is unavoidable and the non-Ricardian set of policy commitments becomes non-credible.

(An optimality principle approach). We may assume that for  $\overline{P}_1 < P_1^F$  the households demand a strictly positive real amount of government bonds. Then, from the definition of equilibrium (condition (ii)) we learn that such a price-contingent individual behavior can not be compatible with the existence of an equilibrium at that  $\overline{P}_1$ . That is, any crisisprice inducing households to demand non-performing bonds will never lead to an observed crisis: it is self-defeated by a households' non-optimal behavior. Notice, however, that a non-optimal household's lending behavior is not sufficient for the government's original fiscal plan to be feasible, in the sense that out of the infinite possible combinations of prices and household's demand for government bonds, only one is compatible with some exogenously targeted  $g_1$  and  $\tau_1$ , namely the one satisfying the following equation

$$\frac{B_1}{\overline{P}_1} = g_1 - \tau_1 + \frac{M_0 + (1 + i_0) B_0 - M_1}{\overline{P}_1}$$
 (22)

Should private lending behavior be rightly described by (22), the government would be in a position to meet all its targets (i.e. to formulate a well-posed consumption demand without abandoning its tax-target and its commitment to redeem the initial debt at its contractual value) at any price provided it satisfies the ongoing private demand for bonds<sup>19</sup>. If, for example,  $\frac{B_1}{P_1}$  falls below the amount in the right side of (22), total resources available for government consumption would be insufficient to go ahead with its original plans. In this sense, (22) is a necessary condition to guarantee the solvency

<sup>&</sup>lt;sup>16</sup>One of Bassetto's (2002) main conclusions is in close connection with this observation: "[...] the strategy I outline above forces the government to increase its taxes in response to a debt crisis; in such an occurrence, not enough resources would be available to pursue the original plan". While he considers the occurrence of a crisis as the outcome of "any (possibly irrational) reason", the crisis I consider here is unambiguously compatible with an individual (contingent) rational behavior.

<sup>&</sup>lt;sup>17</sup>This idea should be clear: I cannot influence the outcome of the market for diamonds by just saying that I want to buy many of them. As the rest of the agents in that market understand that my resources are not in line with my desires, one shouldn't expect any effect in the price of diamonds from my words. Properly speaking, I cannot demand any diamond at a positive price if a do not have a cent and nobody is (perhaps, rationally) lending me. The same is true for the government. The *demand* versus *desire* distinction is clearly captured in Cochrane's words: "you can't double your demand for Porsches, counting on the price to halve." (Cochrane (2003b)).

<sup>&</sup>lt;sup>18</sup>Whether there are alternative government's plans to prevent crisis and the conditions under which may be implemented is the subject of the next sections.

<sup>&</sup>lt;sup>19</sup>As noticed by Cochrane (2003b), "If the consumers *were* willing to lend ever increasing amounts, the government budget constraint must allow them to do so" (italics as in original).

of the government's plans, notwithstanding the fact that it is also a sufficient condition for rendering  $\overline{P}_1$  a non-equilibrium outcome. That is, government solvency requires a particular non-optimal individual behavior. Still, the following are natural questions when analyzing an equilibrium selection device designed by the government: (i) what is the precise mechanism that renders most prices incompatible with the existence of an equilibrium?, and (ii) how is that mechanism implemented by government? In answering the first question (what is the precise mechanism that makes  $\overline{P}_1 < P_1^F$  incompatible with the existence of an equilibrium?), the "optimization principle approach" offers a simple resolution: when the households face a price  $\overline{P}_1 < P_1^F$ , they make a suboptimal lending decision, this behavior being the "true" equilibrium rejection mechanism. However, this argument remains silent with respect to the second query (how is that mechanism implemented by qovernment?). All we know is that a lending decision described by (22) is consistent with the government plans and inconsistent with the existence of equilibrium but we do not know the underlying process leading the household to demand useless bonds when the money-price is "too low" and, critically, we can never deduce a mechanism linking the set of feasible government actions to such a private decision for, as argued above, a rule prescribing a zero demand for government bonds is always feasible, no matter what the price level would happen to be.

Taking as valid this "optimization principle approach" also forces a reconsideration of the cause-effect relationships postulated by the FTPL: what allows the government to formulate a price-invariant demand for goods while insisting on an exogenous level of taxation, a commitment to honor its debts at par and an elastic money-supply rule when faced with a price  $\overline{P}_1 < P_1^F$  is the households' willingness to demand a particular amount of non-performing bonds (dictated by (22)) while the contrary is not true, i.e. it is not possible to combine the set of policy instruments considered in this economy in such a way that the government forces the households to demand worthless bonds. In this precise sense, an equilibrium selection device that rejects some price vectors as potential equilibria by ex ante assuming rather than proving the existence of a conflict between individual rational behavior and prices cannot be considered a policy-based device, but, rather, a psychological-based one.

Finally, the case of a potential crisis-price  $\overline{P}_1$ , such that  $\overline{P}_1 > P_1^F$ , can be handled exploiting the reasonings followed above for the opposite kind of deviations from the fiscalist unique equilibrium, with an added qualification. Now, when considering eventual violations of the transversality condition associated with positive private borrowing (i.e. situations in which the households find a positive demand for their own bonds), one cannot invoke an argument based on sub-optimal household's lending decisions since, indeed, not taking advantage of an opportunity for selling bonds, maturing after death, at a positive price cannot be optimal. Further, if we were to allow the government to passively purchase any amount of bonds supplied by the households paying an strictly positive price, these would optimally supply infinite amounts of debt, thus precluding any economically meaningful equilibrium<sup>20</sup>. In view of these reflections, it seems natural to impose the no-Ponzi games condition  $B_1 \geq 0$  and, similarly,  $B_1^s \geq 0$ , with both inequalities holding for any  $\overline{P}_1$ , so that there is never a market for privately-issued bonds. This

<sup>&</sup>lt;sup>20</sup>This argument can be found in Woodford (1998).

modelling device simplifies the previous analysis to a great extent, as now we do not need to worry about potential violations of the bonds-market clearing condition along which the government tries to implement its original plan by demanding private bonds when faced with an arbitrary  $\overline{P}_1 > P_1^F$ : we are ruling out those violations by assumption and for any price level, not just for one, very much like in the Ricardian case in Proposition 1.

As argued by Weil (2002), imposing  $B_1 \geq 0$  ("consumers are forbidden to die in debt") amounts to impose a feasibility constraint, rather than an optimality constraint. But all we know about feasibility constraints is that they can never be violated, no matter what the price level is or whether such a price can be supported as an equilibrium or not. Hence, a mechanism designed by the government to rule out a crisis-price  $\overline{P}_1 > P_1^F$  based upon the promise to transfer any resources above the (unique) amount compatible with the original fiscal plan to the private sector by purchasing private debt is unfeasible and, hence, non-credible.

In sum, the standard FTPL's interpretation of the government transversality condition (or, equivalently, its intertemporal constraint) as a separate equilibrium condition does not shed any light into the question of how the government may combine its fiscal instruments so as to implement a successful equilibrium selection device able to solve the classical nominal indeterminacy associated with an elastic money supply rule. A government strategy of "keeping on the original plan" in view of a crisis will imply, in most cases, a non-credible "threat", for a well-posed demand function cannot be conceived without respecting an underlying budget set. Altering the government budget set by printing non-performing bonds is not always possible, for someone else must voluntarily stand on the other side of the market ready to rescue a insolvent government plan. However, accepting such a possibility amounts to assume that the households behave in a non-optimal way when nothing precludes them from making the (contingent) optimal decisions. This is not a good axiom to start with for any theory claimed to be an equilibrium selection device founded on a "suitable" design of fiscal rules.

# 4 The limits of well-defined non-Ricardian policies

Using the simple one-period framework presented above, in this section I argue that a suitable re-design of the class of fiscal policies analyzed in the preceding section, which differs from a completely exogenous fiscal policy rule, can effectively rule out some potential equilibria, very much in the spirit of the standard FTPL. However, price and equilibrium uniqueness requires some extra conditions not considered by the standard FTPL. Specifically, a critical (and rather strong) necessary condition for the government to be able to implement an exogenously targeted equilibrium primary surplus while committing to a fully-elastic money-supply rule is that the fiscal plan must involve a non-positive level of seigniorage.

As shown below, implementation of "genuine" non-Ricardian policies (i.e. policies that create a link between prices and private wealth that break the Ricardian Equivalence Theorem) resolves only a part of the indeterminacy problem associated to an elastic supply of money, although this requires a fiscal policy that reacts at off-equilibrium prices, for the presence of non-Ricardian elements is not a sufficient condition for equilibrium uniqueness. Roughly speaking, non-Ricardian fiscal policies solve "half of the problem", and they do so in a way compatible with the arguments given by the advocates of the FTPL, namely, by using fiscal policy to back money with valuable resources collected by the government, i.e. taxes. By contrast, the other half of the problem, which has been systematically neglected, is related to situations in which money injections should back an insufficient amount of taxes. A solution to this latter problem cannot be given by invoking non-Ricardian fiscal-effects. This requires a much more drastic assumption: the government must plan a fiscal policy sufficiently solvent so as to never need new positive, in net terms, monetary injections.

## 4.1 Ruling out government supersolvency

Below I argue that there is a simple policy rule which removes some potential equilibrium price sequences, namely those associated with prices satisfying  $\overline{P}_1 > P_1^F$  in (20), which, otherwise, would force the government to increase its consumption, to reduce taxes or to pay a supersolvency premium on the initial stock of debt (i.e. redeeming bonds above par). I first show the basic insights of this argument by using a simplified version of the one-period model presented above in which the households do not have government bonds at the beginning of the period<sup>21</sup>. Then I extend the analysis to a more general case with a positive initial stock of government debt and, in Section 6, I consider a multiple-period economy.

Consistently with the arguments given in the previous section, I assume that individuals never demand nor supply bonds at the end of the period, i.e.  $B_1 = 0$ , so that the argument does not give any special role to the transversality condition. Let's consider a

 $<sup>^{21}</sup>$ This is consistent with Woodford's (1995) analysis of an economy without initial nominal-bonds holdings.

price level,  $\overline{P}_1$ , such that  $\overline{P}_1 > P_1^F$  (with  $P_1^F$  given in (20) after imposing  $(1 + i_0) B_0 = 0$ ) and assume that the government sets some exogenous targets for  $g_1$  and  $\tau_1$ , labelling these targets as  $\overline{g}_1$  and  $\overline{\tau}_1$ . As (20) holds as an identity, a price  $\overline{P}_1 > P_1^F$  implies that the fiscal targets cannot be attained<sup>22</sup>. For example, if the government insists on consuming  $\overline{g}_1$  regardless of the price level then, given that  $B_1 = 0 \ \forall \overline{P}_1$ , taxes must adjust downwards so as to satisfy (20). The equilibrium associated with  $\overline{P}_1$  would entail the following allocations:

$$g_1 = \overline{g}_1, \quad c_1 = y - \overline{g}_1, \quad M_1(\overline{P}_1) = v'^{-1}[u'(y - g_1)]\overline{P}_1, \quad B_1 = 0$$

with the equilibrium tax,  $\tau_1$ , satisfying

$$\tau_1 = \overline{g}_1 - \frac{M_1(\overline{P}_1) - M_0}{\overline{P}_1}$$

Labelling seigniorage as  $s(P_1)$ , we may express deviations of the actual tax from the target as

$$\overline{\tau}_1 - \tau_1 = s(\overline{P}_1) - s(P_1^F) > 0 \Leftrightarrow \overline{P}_1 > P_1^F$$

This adjustment is plainly consistent with the Ricardian argument in Proposition 1: as the government is receiving more seigniorage,  $\frac{M_1-M_0}{\overline{P}_1}$ , than needed to implement its original consumption plan, its constraint forces a tax reduction, since, this is the only alternative for the government to dispose of its "extra" income from the Ricardian perspective. In what follows, I study the effects of allowing the fiscal branch of the government to consume that extra income through an alternative mechanism, along the lines of the following assumption.

Assumption 1. The fiscal branch of government can participate in the money market buying money and giving consumption goods in exchange.

I do not try to claim any realism in this assumption, for the aim here is just to uncover the necessary conditions under which fiscal policy may reduce the number of potential equilibria. The necessity of this assumption is discussed later. For the moment, it is important to notice that the action by the fiscal authority described in that assumption is not incompatible with the simultaneous central bank's commitment to supply as much money as demanded so it is not at odds with the view of the two authorities, fiscal and monetary, implementing their policy instruments in a decentralized fashion.

Now, suppose that in an eventual situation in which the fiscal branch of the government would receive an amount of real resources (i.e. taxes plus seigniorage) above the one compatible with its exogenous targets, it commits to inelastically supply a positive quantity of goods equal to the "excess" of resources in exchange for any strictly positive amount of dollars. Then any  $\overline{P}_1 > P_1^F$  cannot be part of an equilibrium. The proof of this claim is based on a simple arbitrage argument. Let's suppose that when  $s\left(\overline{P}_1\right) > s\left(P_1^F\right)$  the government keeps its targeted tax, so that its constraint now reads as

$$\overline{g}_1 + w_1 \equiv g_1 = \overline{\tau}_1 + s\left(\overline{P}_1\right) \tag{23}$$

<sup>&</sup>lt;sup>22</sup>Henceforth, I assume that the households behave in a contingent optimal way, i.e. their demands for goods and dollars obey the optimality condition (17) for any price.

where  $w_1$  stands for the excess of budget available for government consumption (with respect to its target  $\overline{g}_1$ ). The commitment to bid  $w_1$  in exchange for any arbitrary strictly positive amount of dollars implies that any individual could buy, e.g., a single dollar from the central bank at a real cost<sup>23</sup> of  $\frac{1}{\overline{P}_1}$  and then supply that dollar in exchange for the (strictly positive) amount of goods supplied by the fiscal authority,  $w_1 + \frac{1}{\overline{P}_1}$ . Such an operation gives that individual a windfall gain of  $w_1$  units of consumption<sup>24</sup>. Clearly, the existence of such a free-lunch opportunity is not compatible with the existence of an equilibrium. Thus, given the commitment to dispose of any eventual excess of seigniorage, by accepting dollars in exchange for it, a new necessary condition must be added to the definition of an equilibrium for this economy: for  $\overline{P}_1$  to be part of an equilibrium,  $\overline{P}_1 - P_1^F$ , cannot be positive.

It is worth noticing that this condition implies that at off-equilibrium prices government consumption,  $\overline{g}_1 + w_1$ , varies with the price, which, in turn, means that available private wealth and, hence, consumption demand, depend on the price, thus creating a link between prices and optimal decisions. In other words, this fiscal price-contingent strategy creates well-defined price-based non-Ricardian wealth effects<sup>25</sup>.

It remains to check that such a government's commitment is a credible one. Clearly, this is the case here, as this commitment only implies that the government makes a "gift" to the households in case these were to pay a level of seigniorage above the (unique) one compatible with the fiscal targets. Further, as such a commitment can be made in a credible way regardless of the particular off-equilibrium price, it provides a "valid" extra condition which helps to rule out some equilibria.

The extension of the previous argument to the case in which the households have a positive stock of government debt at the beginning of the period is straightforward. In that case, the government does not only target a particular value for taxes and government consumption but also aims to redeem its existing debts at their contractual value. According to the arbitrage argument given above, any commitment to give up the excess of seigniorage with respect to the targeted level will be credible, where the targeted level of seigniorage now satisfies,

$$s(P_1^F) \equiv v'^{-1} [u'(y - \overline{g}_1)] - \frac{M_0}{P_1^F} = \frac{(1 + i_0) B_0}{P_1^F} + \overline{g}_1 - \overline{\tau}_1$$

Hence, such a price-contingent fiscal policy will remove any potential equilibrium price in which the government would be forced either to pay a supersolvency-premium on its initial debt or to decrease the level of taxes, i.e. any  $\overline{P}_1 > P_1^F$ . However, notice that

<sup>&</sup>lt;sup>23</sup>Let's assume, for simplicity, that this marginal increase in the total nominal demand for dollars does not have any significant effect on the price level, so that this additional dollar can be purchased at the original price,  $\overline{P}_1$ .

<sup>&</sup>lt;sup>24</sup>Notice that the original "excess of seigniorage" in hands of the fiscal authority,  $s(P_1) - s(P_1^F)$ , is augmented by  $\frac{1}{P_1}$ , i.e. the extra seigniorage received by the sale of the additional dollar used in this arbitrage operation.

<sup>&</sup>lt;sup>25</sup>Notice that the assumption of an economy populated by identical households does not automatically implies the fulfillment of the Ricardian Equivalence Theorem, for such a result breaks down as long as government consumption is not constant (see, e.g., Blanchard and Fischer (1989)).

the wealth effect induced by this fiscal strategy does not arise directly from government bonds, rather, it is associated with the initial dollar holdings: government dollars are net wealth. Not surprisingly, implementation of the non-Ricardian rule described here require a positive initial stock of dollars rather than government bonds.

Why is Assumption 1 necessary in this argument? In this economy, there are four ways for the government to dispose of excess "unwanted" resources: paying a supersolvency premium on the initial stock of debt (i.e. violating its commitment to redeem bonds at par), consuming above the target, reducing taxes below the target and purchasing money paying a price above the market one<sup>26</sup>. Except the last one, any other distribution policy will imply the violation of a fiscal commitment, however without necessarily violating any equilibrium condition, as in the Ricardian case. It is worth noticing that a policy that makes total private outside wealth dependant on the price level need not deliver price uniqueness (e.g. a passive adjustment of government consumption). In this sense breaking the Ricardian Equivalence result is not sufficient to yield a unique equilibrium, for that policy must also generate arbitrage opportunities<sup>27</sup>.

# 4.2 The asymmetry between government default and supersolvency

Can we apply a similar arbitrage argument to rule out equilibrium prices,  $\mathcal{P}_1$ , such that  $\mathcal{P}_1 < P_1^F$ ? The answer depends critically on the sign of  $s\left(P_1^F\right)$ , i.e. the targeted level of seigniorage required to avoid an upward adjustment in the primary surplus.

Firstly, assume that  $s\left(P_1^F\right)>0$ , and, as before, let's first consider the case with zero initial debt, so that a positive level of targeted seigniorage,  $s\left(P_1^F\right)$ , goes in hand with a targeted primary fiscal deficit, i.e.  $\overline{g}_1-\tau_1>0$ . Now, a price  $\overline{P}_1< P_1^F$  cannot be ruled out as a candidate equilibrium price exploiting the non-Ricardian argument outlined above. To see this, notice that when the price  $\overline{P}_1< P_1^F$  and the government consumes  $\overline{g}_1$ , the level of seigniorage actually paid by the households falls below the targeted one, i.e.  $s\left(\overline{P}_1\right)-s\left(P_1^F\right)<0$ . The government can only meet its fiscal targets by inducing individuals to pay the targeted level of seigniorage which, in turn, implies that in face of  $\overline{P}_1$ , it should be able to engineer a policy aimed at depreciating the real value of money, something which is only possible if it manages to increase money supply. In the previous case, for a price  $\overline{P}_1>P_1^F$ , the government was able to "defeat" the ongoing market price,  $\overline{P}_1$ , by inelastically supplying an amount of goods equal to  $s\left(\overline{P}_1\right)-s\left(P_1^F\right)>0$  in exchange for money. Symmetrically, to rule out a market price  $\overline{P}_1< P_1^F$ , the government should be able to inelastically supply an amount of money equal to  $s\left(\overline{P}_1\right)-s\left(\overline{P}_1\right)>0$ , in exchange for consumption goods, where something money and something money

<sup>&</sup>lt;sup>26</sup>In the argument developed here, the government does not commit directly to pay a particular price when redeeming money but, by committing to sell a given amount of consumption goods in exchange of any quantity of dollars, it is indirectly allowing for the possibility of selling goods at a zero net price.

<sup>&</sup>lt;sup>27</sup>This observation resembles one of the conclusions of Cushing (1999). He allows for non-Ricardian households in an OLG model concluding that such a departure is not sufficient for equilibrium uniqueness under a pure interest rate peg.

satisfying (17) for  $P_1^F$  and  $\overline{P}_1$ , respectively. Such a policy announcement, however, is not consistent with the assumed commitment to meet the ongoing demand for nominal balances in a completely elastic fashion. Equivalently, a negative  $w_1$  in (23), holding the target  $\overline{g}_1$ , can only be corrected by increasing  $\tau_1$  or abandoning the monetary rule.

Notice that the inability of the government to put upward pressure on the price level in the present case is not a consequence of the strict separation of competencies between the fiscal authority and the central bank, rather, it is a natural result arising from the specified money supply rule. Even if we think of a central bank fully subordinated to the dictates of a fiscal authority facing financial difficulties, there is not any arrangement capable of providing individuals with the right incentives to pay a higher level of seigniorage under this monetary rule. Therefore, an arbitrary price  $\overline{P}_1 < P_1^F$  can be an equilibrium outcome, with the fiscal authority being forced to either increase taxes or decrease government consumption (or a combination of both). It follows that in this simple setting without initial debt holdings, the government cannot credibly commit to run an arbitrary primary deficit under any circumstance.

As before, extending this nominal indeterminacy result to the more general case in which there is a positive stock of initial government's debts is straightforward. When the standard fiscalist solution in (20) calls for a positive level of seigniorage, i.e.  $s(P_1^F) > 0$ , there is no guarantee that the government will always be able to avoid default if it insists on a particular level of primary surplus.

The reason for the asymmetry between the default and supersolvency cases when  $s(P_1^F) > 0$  is very intuitive. Ruling out default would need a violation of the money supply rule, that is, a critical departure from the context within which the FTPL claims to yield price uniqueness. Ruling out a supersolvency premium only requires a commitment from the government to distribute any excess of seigniorage by purchasing dollars and, as argued before, such a commitment can always be made in a credible way. Moreover, such a fiscal strategy is not incompatible with the commitment to sell as many dollars as demanded at the market price. Put it other way, the ability of the government to rule out supersolvency is possible because for a price such that  $\overline{P}_1 > P_1^F$ , it enjoys a strong financial position, in the sense that it is collecting more resources than needed to go ahead with its original plans. On the other hand, when  $\overline{P}_1 < P_1^F$ , the government faces a weak financial position in that it will be unable to meet its plans at that price. For a particular agent, whether a household or an economic authority, to be able to "defeat" a potential market equilibrium outcome, that agent must be able to generate arbitrage opportunities at the ongoing market price, which is only possible if that agent is willing to face losses in trading with the different assets and goods. The later, using the above terminology, requires that agent to be in a strong financial position. In the next section, I provide further intuition on this asymmetry in the context of a private firm.

# 4.3 The necessity of convertible (non-fiat) money

In this simple one-period model, an obvious way to circumvent the multiplicity of equilibria problem arising when  $s(P_1^F) > 0$  is to further restrict the fiscal-monetary program so

that it satisfies  $s\left(P_1^F\right) \leq 0$ . This non-positive seigniorage condition can then be exploited to rule out prices satisfying  $\overline{P}_1 < P_1^F$ , as potential equilibria, provided the government sets an upper bound to the (negative) volume of seigniorage, as stated in the following assumption<sup>28</sup>.

Assumption 2. The central bank commits to supply any amount of money at the ongoing market price level but does not commit to symmetrically purchase any amount of money at any market price.

Then, only if  $s\left(P_1^F\right) \leq 0$  and Assumptions 1 and 2 hold, it follows that there is unique equilibrium price, which coincides with the fiscalist's standard solution,  $P_1^F$  in (20), provided the government commits to the appropriate price-contingent fiscal rule. The proof for this claim is simple given the previous results. Given that the government can credibly commit to distribute any excess of seigniorage by accepting dollars in exchange for consumption goods, any price  $\overline{P}_1 > P_1^F$  can be ruled out as an equilibrium price. On the other hand, any price  $\overline{P}_1 < P_1^F$  would be associated with a negative level of seigniorage higher, in absolute value, than the targeted one,  $s\left(P_1^F\right)$ . But given Assumption 2, for such a price to be part of an equilibrium, the households must expect a transfer or resources from the government greater than the one compatible with the fiscal objectives. As the government is not obliged to redeem money for an arbitrary amount of consumption goods, such individual's beliefs are not rational and, hence, prices consistent with those beliefs can be ruled out. The necessity of Assumption 2 for this uniqueness result is clear, as well.

Yet, it must recognized that the latter class of deviations from the equilibrium price level (i.e. prices below  $P_1^F$ ) are not corrected by means of a any non-Ricardian wealth effect. Indeed, government consumption may be kept at the targeted value along those deviations. Instead, the driving force to rule out those prices is a pure convertible-asset valuation argument. It is worth providing some intuition about the necessary condition for the above uniqueness result,  $s\left(P_1^F\right) \leq 0$ , relating it to the economic concepts of *convertible* and *fiat* money. Condition  $s\left(P_1^F\right) \leq 0$  implies that, except in the particular case in which  $s(P_1^F) = 0$ , money must be a convertible asset, in the sense that the government is actually retiring money from circulation, i.e.  $M_1(P_1^F) < M_0$ , something which is only possible by giving up some valuable resources (i.e. consumption goods) in exchange for it. Taken literally, that condition forces a reconsideration of the nature of money. When  $s\left(P_{1}^{F}\right)<0$ , money is not longer a purely fiat asset whose real value is exclusively driven by individual expectations on the future acceptability of such an asset by other private agents, but rather, as the fiscal branch of the government repurchases money, its equilibrium real value must also reflect the value of those consumption goods being given in exchange for it. Further, this "dual-value of money" argument also applies in the limiting case in which the government "gives" zero units of consumption goods in exchange for a dollar (i.e.  $s(P_1^F) = 0$  and money is neutral from a fiscal perspective). Notice that price uniqueness in this particular case follows from the assumption of no-speculative

<sup>&</sup>lt;sup>28</sup>Again, the aim here is not to claim the realism of this assumption but rather to impose a minimum set of conditions characterizing the institutional framework within which the FTPL's basic postulates are consistent. As argued later, this assumption is necessary for equilibrium uniqueness.

hyperinflations. If  $\lim_{m\to 0} mv'(m) = 0$ , then speculative hyperinflationary paths cannot be ruled out unless the fiscal authority commits to a price-contingent real backing scheme as in Obstfeld and Rogoff (1983), i.e. even in this case money must be convertible when the Walrasian auctioneer were to dictate a infinite price level. Such a commitment can be readily included in the fiscal authority's set of contingent actions without altering substantially the characterization of the unique equilibrium just described, since that commitment implies that convertibility never takes place in equilibrium.

The role of the necessary conditions for the uniqueness result (Assumptions 1 and 2) is a very intuitive one (as also discussed in the next section). Assumptions 1 and 2, when considered jointly, imply that turning money into a convertible asset is at the full discretion of the government through its fiscal choices. This condition is nothing but a reflection of the high degree of "fiscal dominance" that is necessary for a theory which tries to give fiscal policy a first order importance role in the determination of the price level, as the FTPL does.

The following proposition summarizes the main previous results.

**Proposition 3** In the one-period economy described above, when the government targets a particular primary surplus,  $\overline{\tau}_1 - \overline{g}_1$  and aims to redeem the initial stock of debt at the contractual value, and simultaneously follows a completely elastic money supply rule, the equilibrium price,  $P_1^e$ , satisfies the following conditions:

(i) If Assumption 1 holds and the government commits to inelastically supply any excess of seigniorage,  $s(P_1) - s(P_1^F) > 0$ , in exchange for money, then  $P_1$  cannot be part of an equilibrium. Any equilibrium price must satisfy  $P_1^e \leq P_1^F$ , where  $P_1^F$  is defined as

$$P_1^F = \frac{M_0 + (1 + i_0) B_0}{\overline{\tau}_1 - \overline{g}_1 + v'^{-1} [u'(y - \overline{g}_1)]}$$

- (ii) If the targeted level of seigniorage satisfies  $s(P_1^F) > 0$  and  $P_1^e < P_1^F$ , then the original fiscal-monetary plan cannot be implemented.
- (iii) If Assumptions 1 and 2 hold, the targeted level of seigniorage satisfies  $s\left(P_1^F\right) \leq 0$  and the government commits to inelastically supply any excess of seigniorage in exchange for money, then there is a unique equilibrium price,  $P_1^e$ , satisfying  $P_1^e = P_1^F$ .

The message of this proposition is also useful to understand one of the center claims of the FTPL: the non-necessary cooperative behavior between the fiscal and monetary authorities. As only fiscal plans that involve a non-positive level of seigniorage are credible in face of a commitment to an elastic money supply rule, the fiscal authority does not need any financial help from the central banker, for it is sufficiently solvent so as to never need seigniorage or, perhaps, even to purchase (i.e. destroy) money.

# 5 Learning from the fiscalist stock-analogy

In this section, I discuss to what extent the results of the previous sections force a reconsideration of the fiscalist stock-analogy mentioned earlier. The main claim here is that a strategy based on applying a standard asset valuation approach a la Lucas (1978) to determine the value of money through the "government valuation equation" (13) may be misleading, at least, for two reasons.

The first reason is an obvious one: the RHS of (13) is not a correct measure of the total discounted amount of resources, collected by the government, i.e. that term cannot be understood as the "profits" of the government. To see this, notice that net income gained by the government because of its monopolist production of money, according to the RHS of (13), is given by the so-called inflation-tax term,  $\frac{i_t}{1+i_t}m_t$ , which is not the correct measure of those monopolistic profits, as these correspond to the amount of seigniorage,  $\frac{M_t-M_{t-1}}{P_t}$ . The inflation-tax referred above is a measure of the opportunity cost borne by an individual who accepts to hold a share of his wealth in the form of monetary balances rather than bonds. But such an opportunity cost cannot be thought as of being net income going to the government<sup>29</sup>. Logically, if the RHS (13) is not a correct measure of the firm's profits then the LHS of that equation cannot be thought as of being the number of outstanding shares of the firm, if one is still willing to accept an analogy between a firm and the government. Thus, when the "government valuation equation" is written so as to contain the total discounted value of the total net real government's income in its RHS, what appears on the LHS is the initial stock of government's debts, if any, i.e.,

$$\frac{(1+i_0)B_0}{P_1} = \sum_{t=1}^{\infty} \frac{\tau_t - g_t + \frac{M_t - M_{t-1}}{P_t}}{\prod_{s=1}^{t-1} (1+r_s)}$$
(24)

Equation (24) makes clear that the true "profits" depend on the initial period's price level,  $P_1$ , through the term  $\frac{M_0}{P_1}$ , so the firm's stock-analogy losses its attractiveness here, as we cannot longer assume that total profits are fully independent of the price level.

The second reason is related to a more fundamental observation. The standard FTPL argues that a policy mix resulting in a unique equilibrium sequence of discounted government's total surpluses (i.e. primary surpluses plus seigniorage) will result in a unique equilibrium whenever there is an initial stock of government's nominal obligations. Such an argument would be correct (put aside, for a while, the previous observation on the incorrect measure of total surpluses) whenever the only reason for holding government nominal bonds and, critically, money, is the expectation of redeeming these assets for government-provided valuable resources, i.e., according to the stock-analogy what gives value to money is the amount of resources that the government is going to employ in purchasing money or in distributing dividends on it. This simple and rather strong conclusion should not be surprising since, after all, this is the basic assumption underlying

<sup>&</sup>lt;sup>29</sup>For example, let's consider an infinite-horizon economy in which the level of private consumption, the real interest rate and the supply of money are constant. Constant money supply implies that both seigniorage and inflation (ruling out bubble-solutions) are zero but the inflation-tax will be positive as long as the real interest rate is positive. Identifying this last concept with seigniorage (i.e. the net effective gain of the central bank) only leads to an erroneous calculation of total government's surpluses.

the standard model of equity valuation which the FTPL looks at: the only reason for holding equity, and what gives that equity a real value, is the expectation of receiving profits from the issuer firm via distributed dividends or share repurchases. As the government does not pay any dividend on money (i.e. zero nominal interest on money), the only way in which the stock-analogy can work here is by allowing the government to repurchase money. Not surprisingly, the class of policies studied in the previous section yield the uniqueness result supported by the stock analogy precisely when the policy-mix involves a non-positive level of seigniorage money, i.e. when money effectively gets a share of the government's "profits".

Nevertheless, the fiscalist stock-analogy is still useful to shed some light on several issues discussed before. The asymmetry between government default and supersolvency arising in the "modified" FTPL developed in the previous section is something natural in the context of a firm. To see this, let's write the following equation, reminiscent of the one-period economy government's budget constraint (19),

$$\frac{S_0}{Q_1} = \pi_1 + \frac{S_1}{Q_1}$$

where  $S_0$  and  $S_1$  are the initial and the final number of outstanding shares of the firm, respectively,  $\pi_1$  is the net operational profit of the firm and  $\frac{1}{Q_1}$  is the real value of each share (measured in units of profits). A non-zero terminal portfolio,  $\frac{S_1}{Q_1}$ , can be justified if the share-holders find this firm's paper useful even after the firm has been liquidated, i.e. these S-shares are "tasty" for the share-holders<sup>30</sup>. Let's also assume that the share-holders want to hold a particular terminal portfolio, measured in real terms, i.e.  $\frac{S_1}{Q_1} = \kappa$ , for some given (positive)  $\kappa$ , and that the firm commits to supply as many shares as demanded at the ongoing market price,  $Q_1$ .

Then, if  $\pi_1 > 0$ , the firm offers the investors the possibility of exchanging some shares for profits up to the point at which total profits are fully distributed. When the share-holders behave optimally (i.e. not leaving any valuable profit unexploited), the equilibrium price of each share satisfies

$$\frac{S_0}{Q_1^F} = \pi_1 + \kappa \tag{25}$$

and the number of shares repurchased by the firm is  $S_0 - Q_1^F \kappa$ .

However, if the firm plans to suffer a loss, i.e.  $\pi_1 < 0$ , there is no mechanism to force investors to provide the firm with the required resources, notwithstanding the fact that investors may be willing to buy some new shares, just because they like them. But nothing guarantees that this later form of funding will be enough to allow the firm to go ahead with its plans, even if the RHS of (25) is strictly positive. The firm's commitment to consume no more than  $\pi_1$  rules out any price  $Q_1 > Q_1^F$ , but there are multiple equilibria in which actual losses, in absolute value, are below  $\pi_1$  and the equilibrium price level satisfies  $Q_1 \leq Q_1^F$ .

<sup>&</sup>lt;sup>30</sup>For a similar "tasty-share" story, see, e.g., Marimón (2001).

We can extend this simple analogy by introducing "nominal bonds", that is, a new type of shares that are not perceived as intrinsically useful by the holders and that are denominated in terms of the tasty ones (call them N-shares). Let's focus on the case in which the firm plans a positive level of operational profits, i.e.  $\pi_1 > 0$ , and tries to repurchase the outstanding no-tasty shares,  $N_0$ , at a price equal to  $Q_1^F$  in the following equation

 $\frac{S_0 + N_0}{O_1^F} = \pi_1 + \kappa$ 

Also, let's assume that  $S_0$ ,  $N_0$ ,  $\pi_1$  and  $\kappa$  are such that the firm needs to sell new S-shares to meet its plans, i.e.  $S_1 - S_0 > 0$ . Then, at any price  $Q_1 > Q_1^F$ , the firm is not distributing all its profits even after having repurchased the entire initial stock of N-shares. A commitment to distribute any remaining profits via repurchases of S-shares renders such a price  $Q_1 > Q_1^F$  a non-equilibrium outcome: not accepting such a repurchase offer amounts to give up a profitable opportunity for the investors. On the other hand, if  $Q_1 < Q_1^F$ , new net capital injections, as measured by  $\kappa - \frac{S_0}{Q_1}$ , are not sufficient to repurchase the outstanding  $N_0$  shares at the ongoing price  $Q_1$ . What can the firm do? It cannot directly sell more newly printed S-shares, given the commitment to a completely elastic supply rule. It cannot increase its operational profits, as they are assumed to be given exogenously. There is nothing the firm can do to provide investors with the right incentives to purchase new S-shares, promising, at the same time, that this extra income will be rebated to those same investors through a higher redemption value for the N-shares and, as a result, multiple equilibria along which each N-share takes a value below a S-share cannot be ruled out.

An interesting example shows up when assuming that the investors do not find Sshares useful per se anymore (i.e.  $\kappa = 0 \ \forall Q_1$ ). Under this assumption, the initial claims on the firm's profits,  $S_0$  and  $N_0$ , may always have a unique equilibrium common value since nothing may preclude the firm from committing to distribute its profits across the different types of shares evenly (i.e. redeeming both types of shares at the same value). Of course, if the only outstanding claims are N-shares  $(S_0 = 0)$ , there will be a unique equilibrium price for those shares and, naturally, the price for a security that is not traded at all (S-shares) is not defined. Thus, what makes the problem of pricing S-shares a non-trivial one is the possibility of an existing demand for these securities which is not exclusively driven by the investors' expectations on the firm's stream of profits or losses (in this simple example, the existence of a positive portfolio  $\kappa$ , and in a monetary economy, the public's perception of money as an object which provides utility per se, helps to overcome a cash-in-advance constraint or reduces the cost of making transactions). In such a case, a new kind of profit, different form the operational one, emerges, for net sales of these shares increase the firm's net worth. Further, when the firm commits to an elastic supply of S-shares, the amount of this later form of profit will not be uniquely determined, unless the firm actually uses a portion of its operational profits to purchase those shares, in which case, in equilibrium, S-shares are not longer an additional source of (indeterminate) profits but a destination in the distribution process of the exogenous (and uniquely determined) operational gains. In the same fashion, what makes the problem of pricing money a non-trivial one and, potentially, a "difficult" one is the possibility that private individuals find money useful even if it is well understood that the government

will never redeem that money for consumption goods, i.e. the possibility of money being a fiat asset.

Also, as the examples above show, the problem of default is linked to the existence of an underlying commitment to redeem the entire stock of N-shares at the price of each S-share. In this sense, default is a true possibility for certain firm's claims as it is a true possibility for government dollar-denominated bonds. Were this commitment be absent (as in Bassetto (2002)), a concern on a possible debt-default would no longer be justified.

Comparing the above examples with the economy analyzed in the previous section yields, indeed, an interesting analogy. In particular, Assumptions 1 and 2 presented above, which are necessary for equilibrium uniqueness, can be also seen as the necessary elements to go from the standard institutional framework of a firm to the one described in the preceding section. Notice that Assumption 2 implies that money, while being potentially a convertible asset, is just a residual claim to the eventual government surpluses, that is, there is nothing forcing the government to redeem any arbitrary quantity of dollars at any arbitrary price, very much like the S-shares above. Assumption 1 plays a crucial role in removing potential equilibria in which the government, having committed to an exogenous primary surplus, would be forced to redeem the existing debt at a value above the contractual one. Indeed, this assumption can be seen as of capturing the autonomy of a solvent firm in deciding how to distribute its profits across different classes of shares (money and bonds).

Thus, once we assume that a firm's total profits may depend on the seigniorage gained by the sales of tasty shares, there is an analogy between the mechanism of determination of the equilibrium value of those tasty shares and the one governing the value of government-issued money when a correctly specified (i.e. credible) fiscal-monetary plan that involves a non-positive level of seigniorage is at work.

# 6 A multiple-period economy

In this section I extend the arguments developed before to an economy with multiple periods. This scenario allows for an explicit consideration of the nominal rate as a policy instrument. Before developing the formal arguments, it is useful to briefly discuss the nature of the commitment associated to an interest rate peg and its influence on private expectations. Proposition 1 states that, in a perfect-foresight economy, under a Ricardian policy the ability of the government to commit to a particular nominal return paid by its debt regardless of the initial price level implies that, while the total present value of surplus depends on the particular realization of  $P_1$ , the sequence of equilibrium inflation rates is independent of  $P_1$ . Thus, at least regarding the determination of inflation, Ricardian policies in a perfect foresight setting do not pose important difficulties. However, as stressed by Sargent and Wallace (1975), and more recently by Woodford (2003), as we consider a scenario with uncertainty, a policy of pegging the nominal interest rate will involve price-indeterminacy at each date, and hence, the *actual* inflation rate will be indeterminate even if it is unique *in expectation*. The basis for a formal argument is simple. Let's consider a linear version of a stochastic Fisher equation<sup>31</sup>,

$$i_t = \widetilde{r}_t + E_t \pi_{t+1} \tag{26}$$

where  $\tilde{r}_t$  is the targeted real return paid by a bond issued at t and maturing at t+1, assumed to depend upon some fundamentals (e.g. preferences, endowments and government consumption) and  $E_t \pi_{t+1}$  is the expectation as of t of the inflation rate prevailing between t and t+1. For given  $i_t$  (assumed to be set directly by the government) and  $\tilde{r}_t$ , (26) can be solved for a unique  $E_t \pi_{t+1}$ . However, (26) does not have a unique finite solution for the stochastic process  $\{\pi_{t+1}\}$ , as any finite  $\pi_{t+1}$  satisfying

$$\pi_{t+1} = i_t - \widetilde{r}_t + \psi_{t+1} \tag{27}$$

is a solution of (26), provided  $\psi_{t+1}$  is not forecastable at t, i.e.  $E_t\psi_{t+1}=0$ . We can not say much about  $\psi_{t+1}$ , apart from this latter condition, as a non-zero  $\psi_{t+1}$  could arise as the result of an underlying process linking prices to fundamentals but critically, as stressed by Woodford (2003; sec. 2.2.1), also as a pure speculative component in the process generating prices (sunspot states), totally unrelated to fundamentals. In a Ricardian world, the distinction of fundamental vs. non-fundamental elements in the solution for the realized inflation rate (i.e. in the price  $P_{t+1}$ ) is unimportant regarding the private rational expectation formed one period earlier about  $P_{t+1}$ , for the public's perception that the government will adjust its fiscal instruments so as to honor its maturing debt at the contractual value implies that for a given  $\tilde{r}_t$ ,  $E_t\psi_{t+1}=0$  must hold always. In this sense, fiscal commitment rules out arbitrary self-fulfilled prophecies on  $E_t\psi_{t+1}$ , and hence a policy of fixing the nominal interest rate paid by government bonds, coupled with a Ricardian fiscal rule will induce a unique inflation expectation. If, on the other hand, the public understands that such a fiscal response at t+1 will not be observed, say because the government announces a sequence of primary surpluses totally independent of the price

<sup>&</sup>lt;sup>31</sup>A detailed derivation of a similar equilibrium condition as a local approximation in an optimizing model can be found in Woodford (2003, C. 2).

level, the previous reasoning will not apply. Under such a fiscal rule, the announcement of a particular nominal interest may not be credible and arbitrary expectations on  $\psi_{t+1}$  consistent with individual optimization can not be ruled out in advance, even if there is not any other source of uncertainty beyond the random process that produces a particular expectation on  $\psi_{t+1}$ . In what follows I argue that such a possibility must not be disregarded, unless the fiscal plan involves the clear commitment to devote a fraction of the targeted primary surpluses to repurchase a fraction of the beginning-of-period stock of money, as in the one-period economy studied before.

To prove this claim in the simplest possible environment, let's consider a two-period economy, similar in every other respect to the one presented before. Along the lines of the non-Ricardian policy described in Section 2, it is assumed that the government commits to implement a sequence of price-invariant primary surpluses. The government also announces the nominal interest rate prevailing between periods 1 and 2,  $i_1$ . Since the aim here is to show that such a plan involving sequences of exogenously targeted primary surpluses and nominal interest rates may not be always credible, it is convenient to distinguish between government's targeted values for its policy instruments and households' beliefs on those values. For this purpose I introduce a new notation to accommodate the eventual differences between government-announced and household-expected nominal interest rates, the former being denoted by  $i_1^a$  and the later by  $i_1^e$ . Thus, the difference between these rates can be understood as the presence of a sunspot component in the expectation on the actual rate (i.e. using the notation above, I am considering here  $E_t\psi_{t+1} = \psi_{t+1} \neq 0$ , where the equality is imposed to preserve the certainty-hypothesis<sup>32</sup>).

The household's budget constraints are now given by (15) and

$$c_2 \le y_2 - \tau_2 + \frac{(1+i_1)B_1 + M_1 - M_2 - B_2}{P_2}$$

where  $i_1$  is the observed (actual) interest rate paid by government bonds. As before, it is assumed that the household follows a bonds demand rule in the terminal period which is independent of the observed price level, i.e.  $B_2 = 0$  for any  $P_2$ . Following a similar argument, the households' demand for bonds in the first period is assumed to be optimal given their beliefs about the government's resources backing its debt, i.e.,

$$\frac{B_1(i_1^e)}{P_1} = \frac{1}{1+r_1} \left[ \overline{\tau}_2 - \overline{g}_2 + m_2 - \frac{m_1(i_1^e)P_1}{P_2} \right]$$
 (28)

The LHS of (28) represents the demand for government bonds expressed in real terms and the RHS measures the discounted value of the resources available for debt redemption in the following period, i.e. the primary surplus plus seigniorage. The discount factor is defined, according to (6), as  $\frac{1}{1+r_1} = \beta \frac{u'(c_2)}{u'(c_1)}$ . In (28) I am representing the demand for real balances in the first period as  $m_1(i_1^e)$  to explicitly account for the fact that it depends directly on the nominal interest rate the individuals expect (see (5)). As before,  $m_2$ , being the demand for real balances in the terminal period, satisfies a first order condition similar to (17).

 $<sup>^{32}</sup>$ Considering a non-degenerate distribution function for  $\psi_{t+1}$  will not alter the argument.

First, let's suppose that  $i_1^e > i_1^a$ , i.e. the households expect a nominal return on bonds higher than the one announced by the government. Then, using (6) we can express the level of seigniorage in period 2 for an interest rate  $i_1^e$  as

$$s_2(i_1^e) = m_2 - \frac{1 + r_1}{1 + i_1^e} m_1(i_1^e)$$

In order to learn whether those beliefs can be self-confirmed in equilibrium, I consider a vector of private consumption demand  $\{c_1, c_2\}$  compatible with the goods-market clearing condition. Then, for  $i_1^e > i_1^e$ , the following inequality must hold

$$s_2(i_1^e) > s_2(i_1^a)$$
 (29)

In face of (29), the government may commit to follow a policy of rebating the excess of resources,  $s_2(i_1^e) - s_2(i_1^a)$ , giving it in exchange for any arbitrarily positive amount of dollars (Assumption 1), thus creating an arbitrage opportunity which cannot be compatible with an equilibrium. As individuals anticipate that the government will distribute the "excess of profits" in exchange for dollars rather than for bonds, the expected effective return for bonds cannot be above the announced one. That is, if the government is determined to defend the announced nominal rate there cannot be any equilibrium with  $i_1^e > i_1^a$ .

Second, let's consider the case in which  $i_1^e < i_1^a$ , so that  $s_2(i_1^e) < s_2(i_1^a)$ . Then, as long as  $s_2(i_1^a) > s_2(i_1^e) > 0$ , we cannot rule out equilibria in which the government defaults. Given that the government commits to run an exogenous primary surplus, individuals correctly anticipate that the level of seigniorage in period 2 will be insufficient to redeem the stock of bonds issued one period before at their contractual value, increasing the relative demand for real balances and decreasing real lending to the government. Notice that in this economy the fact that (partial) nominal default is anticipated does not necessarily eliminate the incentives of the household to lend real resources to the government by purchasing bonds in period 1 at the price of one dollar. Formally, the relevant condition governing the optimal household's decision on how much to lend can be discomposed into two separate decisions. First, the decision of whether to lend or not at all. The household will be willing to lend some valuable resources, in a positive and finite amount, if the following first order condition holds,

$$(1+i_1^e)\frac{P_1}{P_2} = \frac{u'(c_1)}{\beta u'(c_2)}$$
(30)

Therefore, as long as the expected inflation rate,  $\frac{P_2}{P_1}$ , satisfies (30), the fact that  $i_1^e < i_1^a$  is irrelevant for this decision. All that matters is that each unit of consumption goods invested in bonds at period 1 yields the required real return. Thus, as discussed above, the government's lack of commitment to respond to deviations of the effective nominal rate from the announced one allows for beliefs on expected inflation to become self-confirmed in equilibrium<sup>33</sup>. Second, as an expected positive rate of default is associated in equilibrium

<sup>&</sup>lt;sup>33</sup>The trade-off between a standard Ricardian policy and the exogenous fiscal policy considered here becomes clear. The former is successful in pinning down inflation expectations, however at the cost of leaving taxes indeterminate. The opposite is true for the latter.

with a lower rate of inflation (see (30)) rational individuals will reduce the real amount of debt purchased in period 1. But this reduction is not a direct consequence of the rational expectation of nominal default, instead it is a reflection of the fact that seigniorage in period 2 and, hence, the amount of resources available to payback government's debt falls.

Finally, when  $i_1^e < i_1^a$  and  $s_2(i_1^e) < s_2(i_1^a) \le 0$ , under Assumptions 1 and 2 and an excess seigniorage-rebating policy,  $i_1^e$  cannot be a rational belief, so that the unique equilibrium effective nominal interest is the announced one,  $i_1^a$ , i.e. the standard fiscalist non-default assumption holds. Thus, the necessary condition for the whole non-Ricardian program to be credible in the second period derived here,  $s_2(i_1^a) \leq 0$ , is similar to the one derived before for the one-period economy (Proposition 3 (iii)). Notice, however, that the credibility of the whole fiscal-monetary program requires condition  $s_t(i_{t-1}^a) \leq 0$  to hold for any t, and not just for the (eventual) terminal period. In the two-period economy analyzed here, we must impose  $s_1(i_1^a) \leq 0$  and  $s_2(i_1^a) \leq 0$  in order to obtain a unique equilibrium. For example, if  $s_2(i_1^a) \leq 0$  holds, so that  $i_1^e = i_1^a$ , we learn from (28) that the household's demand for government bonds at the end of period 1, in equilibrium, is unique, given the credible government announcements for period 2. It follows that the problem faced by the household in period 1 is identical to the one faced in the oneperiod economy, as we only need to augment the end-of-period 1 household's equilibrium portfolio with that extra debt-term<sup>34</sup>. Thus, the set of fiscal announcements for period 1 (exogenously targeted primary surplus and redemption of the initial debt, if any, at par) will only be credible if  $s_1 \leq 0$ .

The recursive nature of the optimization problem faced the household implies that we can extend this argument to economies with longer (and infinite) horizons in a straightforward way. The following proposition summarizes these findings.

**Proposition 4** A fiscal-monetary plan in which the government commits to maintain exogenously targeted sequences of taxes, government consumption and nominal interest rates is only credible if Assumptions 1 and 2 hold and, at the targeted policy-instruments sequences, the following condition holds for every  $t \geq 1$ 

$$M_t < M_{t-1} \tag{31}$$

Corollary When Assumptions 1 and 2 and condition (31) hold and the government can commit to maintain its targets (i.e. the announced fiscal-monetary program is credible), the equilibrium is unique.

<sup>&</sup>lt;sup>34</sup>The fact that the expected amount of resources backing government bonds issued at the end of 1 is given uniquely implies that households' demand for bonds cannot differ from that quantity when behaving optimally.

# 7 Concluding comments

In this paper, I first examine the off-equilibrium behavior of the economic agents prescribed by the FTPL, showing that the way in which non-Ricardian policies are usually defined by the advocates of the FTPL involves a non-credible commitment by the government whenever the assumed fiscal-monetary program contains the announcement for maintaining a completely exogenous sequence of government consumption and a fully elastic money supply rule consistent with the central bank announcing an arbitrary nominal interest rate<sup>35</sup>, thus showing that the standard FTPL is never a policy-based equilibrium selection device. The main novelty of this criticism is that it is based on the same core assumptions maintained by the proponents of this theory: there is a positive stock of government-issued assets at the beginning of the history owned by the households, flow of funds constraints must be respected in every contingency, although transversality conditions may be violated at off-equilibrium prices.

Then, using a standard Walrasian model in which money is introduced as an argument in the household's utility function, I identify the minimum set of conditions guarantying the credibility of a fiscalist policy close in spirit to the standard FTPL's non-Ricardian policies. Firstly, I show that when the fiscal authority targets a particular (exogenously determined) sequence of primary surpluses, any equilibrium in which the government would be forced to increase the transfer of resources to the private sector above the targeted level can be ruled out only by assuming that the fiscal authority may participate in the money market purchasing dollars in exchange for real consumption goods. This assumption can then be incorporated into a fiscal strategy that breaks down the Ricardian Equivalence Theorem. Further, it is shown that such a fiscal strategy must necessarily be price-contingent for it to be credible. Secondly, I argue that the previous equilibriumrejection device is not always implementable to symmetrically rule out equilibria in which the government is forced to reduce the transfer of resources to the private sector below the targeted level, as such a device would be inconsistent with the assumed monetary policy rule. Thirdly, I show that a fiscal-monetary plan resulting in a unique equilibrium under a nominal interest rate peg is only credible if that plan involves a non-positive level of seigniorage. That is, the main arguments of this modified FTPL are only valid for non-fiat money. This is one of the most interesting conclusions of this paper.

Finally, I argue that the so-called stock-analogy exploited by some advocates of the FTPL, according to which there exist fiscal strategies that turn the government intertemporal constraint into a "government valuation equation" requires an important qualification for it to be a useful analogy to explain how the equilibrium value of money is determined, namely, that money must be a convertible asset, a result which is plainly consistent with the above arguments about the credibility of a non-Ricardian policy.

 $<sup>^{35}</sup>$ The terms "exogenous" and "arbitrary" are used here in the precise sense of the corresponding policy-instruments not being chosen according to a Ricardian rule.

#### References

- Afonso, A. 2002. "Disturbing the Fiscal Theory of the Price Level: Can It Fit the EU-15?", mimeo, Universidade Técnica de Lisboa.
- Andrés, J., F. Ballabriga and J. Vallés. 2002. "Non-Ricardian Fiscal Policies in an Open Monetary Union", *Economic Papers*, European Commission.
- Arce, O. 2004. "The Fiscal Theory of the Price Level under a Price-Invariant Nominal Money Supply Rule: What Went Wrong?", mimeo, LSE..
- Barro, R. J. 1974. "Are Government Bonds Net Wealth?", *Journal of Political Economy*, 82: 1095-1117.
- Bassetto, M. 2002. "A Game-Theoretic View of the Fiscal Theory of the Price Level", *Econometrica*, 70: 2167-2195.
- Benhabib, J., S. Schmitt-Grohe and M. Uribe. 2002. "Avoiding Liquidity Traps", Journal of Political Economy, 110: 535-563.
- Bergin, P.R. 2000. "Fiscal Solvency and Price Level Determination in a Monetary Union", *Journal of Monetary Economics*, 45: 37-53.
- Blanchard, O. J., and S. Fischer. 1989. *Lectures on Macroeconomics*, Cambridge, MA: The MIT Press.
- Buiter, W. 1999. "The Fallacy of the Fiscal Theory of the Price Level", NBER Working Paper 7302.
- Buiter, W. 2002. "The Fiscal Theory of the Price Level: A Critique", *Economic Journal*, 112 (July): 459-480.
- Canzoneri, M. B., R. E. Cumby, and B. T. Diba. 2001. "Is the Price Level Determined by the Needs of Fiscal Solvency?", *American Economic Review* 91: 1221-1238.
- Christiano, L. J. and Fitzgerald, T. J. 2000. "Understanding the Fiscal Theory of the Price Level", Federal Reserve Bank of Cleveland Economic Review 36: 2-38.
- Cochrane, J. 1998. "A Frictionless View of U.S. Inflation", in Ben Bernanke and Julio Rotemberg, eds., *NBER Macroeconomics Annual* (323-384). Cambridge, MA: The MIT Press.
- Cochrane, J. 2001. "Long-Term Debt and Optimal Policy in the Fiscal Theory of the Price Level", *Econometrica*, 69: 69-116.
- Cochrane, J. 2003a. "Fiscal Foundations of Monetary Regimes", mimeo, University of Chicago.
  - Cochrane, J. 2003b. "Money as Stock", mimeo, University of Chicago.
  - Corsetti, G., and Mackowiak, B. 2003. "A Fiscal Perspective on Currency Crises and

- "Original Sin", mimeo, European University Institute.
- Cushing, M. J. 1999. "The Indeterminacy of Prices under Interest Rate Pegging: The non-Ricardian Case", *Journal of Monetary Economics*, 44: 131-148.
- Daniel, B. C. 2001a. "The Fiscal Theory of the Price Level in an Open Economy", *Journal of Monetary Economics*, 48: 293-308.
- Daniel, B. C. 2001b. "A Fiscal Theory of Currency Crisis", *International Economic Review*, 42: 969-998.
- Daniel, B. C. 2003. "Fiscal Policy, Price Surprises, and Inflation", mimeo, University at Albany (SUNY).
- Daniel, B. C. 2004. "Monetary Policy and Inflation in a Fiscal Theory Regime", mimeo, University at Albany (SUNY).
- Davig, T., E. M. Leeper, and H. Chung. 2004. "Monetary and Fiscal Policy Switching", NBER Working Paper 10362.
- Dupor, B. 2000. "Exchange Rates and the Fiscal Theory of the Price Level", *Journal of Monetary Economics*, 45: 613-630.
- Gordon, D. B. and E. M. Leeper. 2002. "The Price Level, the Quantity Theory of Money, and the Fiscal Theory of the Price Level", NBER, Working Paper 9084.
- Kocherlakota, N. R. and C. Phelan. 1999. "Explaining the Fiscal Theory of the Price Level", Federal Reserve Bank of Minneapolis Quarterly Review, 23: 14-23.
- Leeper, E. 1991. "Equilibria under Active and Passive Monetary Policies", *Journal of Monetary Economics*, 27: 129-147.
- Loyo, E. 2001. "Tight Money Paradox on the Loose: A Fiscalist Hyperinflation", mimeo, JFK School of Government, Harvard University.
- Lucas, R. J. 1978. "Asset Prices in an Exchange Economy", *Econometrica*, 46: 1426-1445.
- Mackowiak, B. 2003. "Macroeconomic Regime Switching and Speculative Attacks", mimeo, Humbolt University.
- Marimón, R. 2001. "The Fiscal Theory of Money as an Unorthodox Financial Theory of the Firm", in Axel Leijonhufvud, ed. *Monetary Theory as a Basis for Monetary Policy*, New York: Palgrave.
- McCallum, B. T. 2001. "Indeterminacy, Bubbles, and the Fiscal Theory of the Price Level", *Journal of Monetary Economics*, 47: 19-30.
- McCallum, B. T. 1999. "Theoretical Issues Pertaining to Monetary Unions", NBER Working Paper 7393.
  - Nicolini, J. P. 1996. "Ruling Out Speculative Hyperinflations. The Role of the Gov-

- ernment", Journal of Economic Dynamics and Control, 20: 791-809.
- Niepelt, D. 2002. "The Fiscal Myth of the Price Level", mimeo, IIES, Stockholm University.
- Niepelt, D. 2004. "The Fiscal Myth of the Price Level", Quarterly Journal of Economics, 119: 277-300.
- Obstfeld, M. and K. Rogoff. 1983. "Speculative Hyperinflations in Maximizing Models: Can We Rule Them Out?", *Journal of Political Economy*, 91: 675-687.
  - Sala, L. 2002. "Testing the Fiscal Theory of the Price Level", mimeo, ECARES.
- Sargent, T. J. and N. Wallace. 1975. "Rational' Expectations, the Optimal Monetary Instrument, and the Optimal Money Supply Rule", *Journal of Political Economy*, 83: 241-254.
- Sargent, T. J. and N. Wallace. 1981. "Some Unpleasant Monetarist Arithmetic", Federal Reserve bank of Minneapolis Quarterly Review, (Fall).
- Sims, C. A. 1994. "A Simple Model for Study of the Determination of the Price Level and the Interaction of Monetary and Fiscal Policy", *Economic Theory*, 43: 381-399.
- Sims, C. A. 1997. "Fiscal Foundations of Price Stability in Open Economies", mimeo, Princeton University.
- Sims, C. A. 1999. "Domestic Currency Denominated Government Debt as Equity in the Primary Surplus", mimeo, Princeton University.
- Sims, C. A. 2002. "Fiscal Consequences for Mexico of Adopting the Dollar", mimeo, Princeton University.
  - Weil, P. 2002. "Reflections on the Fiscal Theory of the Price Level", mimeo, ECARES.
- Woodford, M. 1994. "Monetary Policy and Price Level Determinacy in a Cash-in-Advance Economy", *Economic Theory*, 43: 345-380.
- Woodford, M. 1995. "Price Level Determinacy without Control of a Monetary Aggregate", Carnegie-Rochester Conference Series on Public Policy, 43: 1-46.
  - Woodford, M.1998. "Public Debt and the Price Level", mimeo, Princeton University.
- Woodford, M. 2001. "Fiscal Requirements for Price Stability", *Journal of Money, Credit, and Banking*, 33: 669-728.
- Woodford, M. 2003. *Interest and Prices: Foundations of a Theory of Monetary Policy*, Princeton, NJ: Princeton University Press.

#### **BANCO DE ESPAÑA PUBLICATIONS**

#### WORKING PAPERS<sup>1</sup>

- 0216 ALICIA GARCÍA HERRERO, JAVIER SANTILLÁN, SONSOLES GALLEGO, LUCÍA CUADRO AND CARLOS EGEA: Latin American Financial Development in Perspective.
- 0217 SONSOLES GALLEGO, ALICIA GARCÍA HERRERO AND JESÚS SAURINA: The Asian and European Banking Systems: The case of Spain in the Quest for Development and Stability.
- 0218 JOSÉ RAMÓN MARTÍNEZ RESANO AND LILIANA TOLEDO FALCÓN: Futuros sobre acciones: Demanda e implicaciones sobre los mercados de renta variable.
- 0219 JOSÉ MANUEL CAMPA AND JOSÉ M. GONZÁLEZ MÍNGUEZ: Differences in exchange Rate Pass-Through in the Furo Area
- 0220 LUIS JULIÁN ÁLVAREZ GONZÁLEZ AND JAVIER JAREÑO MORAGO: ISIS, Un indicador sintético integral de los servicios de mercado.
- 0221 MARCO HOEBERICHTS: The Credibility of Central Bank Announcements.
- 0222 KLAUS DESMET: Asymmetric Shocks, Risk Sharing, and the Latter Mundell.
- 0223 JOSÉ MANUEL CAMPA AND IGNACIO HERNANDO: Value creation in European M&As.
- 0224 JUAN AYUSO HUERTAS, DANIEL PÉREZ CID AND JESÚS SAURINA SALAS: Are capital buffers pro-cyclical? Evidence from Spanish panel data.
- 0225 ANDREW BENITO: Does job insecurity affect household consumption?
- 0226 ANDREW BENITO: Financial pressure, monetary policy effects and inventory adjustment by UK and Spanish firms.
- 0227 ANDREW BENITO AND IGNACIO HERNANDO: Extricate: Financial Pressure and Firm Behaviour in Spain.
- 0228 ANA DEL RÍO: El endeudamiento de los hogares españoles.
- 0229 GABRIEL PÉREZ QUIRÓS AND JORGE SICILIA: Is the European Central Bank (and the United States Federal Reserve) predictable?
- 0301 JAVIER ANDRÉS, EVA ORTEGA AND JAVIER VALLÉS: Market structure and inflation differentials in the European Monetary Union.
- 0302 JORDI GALÍ, MARK GERTLER AND J. DAVID LÓPEZ-SALIDO: The euro area inefficiency gap.
- 0303 ANDREW BENITO: The incidence and persistence of dividend omissions by Spanish firms.
- 0304 JUAN AYUSO AND FERNANDO RESTOY: House prices and rents: an equilibrium asset pricing approach.
- 0305 EVA ORTEGA: Persistent inflation differentials in Europe.
- 0306 PEDRO PABLO ÁLVAREZ LOIS: Capacity utilization and monetary policy.
- 0307 JORGE MARTÍNEZ PAGÉS AND LUIS ÁNGEL MAZA: Analysis of house prices in Spain. (The Spanish original of this publication has the same number).
- 0308 CLAUDIO MICHELACCI AND DAVID LÓPEZ-SALIDO: Technology shocks and job flows.
- 0309 ENRIQUE ALBEROLA: Misalignment, liabilities dollarization and exchange rate adjustment in Latin America.
- 0310 ANDREW BENITO: The capital structure decisions of firms: is there a pecking order?
- 0311 FRANCISCO DE CASTRO: The macroeconomic effects of fiscal policy in Spain.
- 0312 ANDREW BENITO AND IGNACIO HERNANDO: Labour demand, flexible contracts and financial factors: new evidence from Spain.
- 0313 GABRIEL PÉREZ QUIRÓS AND HUGO RODRÍGUEZ MENDIZÁBAL: The daily market for funds in Europe: what has changed with the EMU?
- 0314 JAVIER ANDRÉS AND RAFAEL DOMÉNECH: Automatic stabilizers, fiscal rules and macroeconomic stability
- 0315 ALICIA GARCÍA HERRERO AND PEDRO DEL RÍO: Financial stability and the design of monetary policy.
- 0316 JUAN CARLOS BERGANZA, ROBERTO CHANG AND ALICIA GARCÍA HERRERO: Balance sheet effects and the country risk premium: an empirical investigation.
- 0317 ANTONIO DÍEZ DE LOS RÍOS AND ALICIA GARCÍA HERRERO: Contagion and portfolio shift in emerging countries' sovereign bonds.
- 0318 RAFAEL GÓMEZ AND PABLO HERNÁNDEZ DE COS: Demographic maturity and economic performance: the effect of demographic transitions on per capita GDP growth.
- 0319 IGNACIO HERNANDO AND CARMEN MARTÍNEZ-CARRASCAL: The impact of financial variables on firms' real decisions: evidence from Spanish firm-level data.

<sup>1.</sup> Previously published Working Papers are listed in the Banco de España publications calalogue.

- 0320 JORDI GALÍ, J. DAVID LÓPEZ-SALIDO AND JAVIER VALLÉS: Rule-of-thumb consumers and the design of interest rate rules.
- 0321 JORDI GALÍ, J. DAVID LÓPEZ-SALIDO AND JAVIER VALLÉS: Understanding the effects of government spending on consumption.
- 0322 ANA BUISÁN AND JUAN CARLOS CABALLERO: Análisis comparado de la demanda de exportación de manufacturas en los países de la UEM.
- 0401 ROBERTO BLANCO, SIMON BRENNAN AND IAN W. MARSH: An empirical analysis of the dynamic relationship between investment grade bonds and credit default swaps.
- 0402 ENRIQUE ALBEROLA AND LUIS MOLINA: What does really discipline fiscal policy in emerging markets? The role and dynamics of exchange rate regimes.
- 0403 PABLO BURRIEL-LLOMBART: An economic analysis of education externalities in the matching process of UK regions (1992-1999).
- 0404 FABIO CANOVA, MATTEO CICCARELLI AND EVA ORTEGA: Similarities and convergence in G-7 cycles.
- 0405 ENRIQUE ALBEROLA, HUMBERTO LÓPEZ AND LUIS SERVÉN: Tango with the gringo: the hard peg and real misalignment in Argentina.
- 0406 ANA BUISÁN, JUAN CARLOS CABALLERO AND NOELIA JIMÉNEZ: Determinación de las exportaciones de manufacturas en los países de la UEM a partir de un modelo de oferta-demanda.
- 0407 VÍTOR GASPAR, GABRIEL PÉREZ QUIRÓS AND HUGO RODRÍGUEZ MENDIZÁBAL: Interest rate determination in the interbank market.
- 0408 MÁXIMO CAMACHO, GABRIEL PÉREZ-QUIRÓS AND LORENA SAIZ: Are European business cycles close enough to be just one?
- 0409 JAVIER ANDRÉS, J. DAVID LÓPEZ-SALIDO AND EDWARD NELSON: Tobin's imperfect assets substitution in optimizing general equilibrium.
- 0410 A. BUISÁN, J. C. CABALLERO, J. M. CAMPA AND N. JIMÉNEZ: La importancia de la histéresis en las exportaciones de manufacturas de los países de la UEM.
- 0411 ANDREW BENITO, FRANCISCO JAVIER DELGADO AND JORGE MARTÍNEZ PAGÉS: A synthetic indicator of financial pressure for Spanish firms.
- 0412 JAVIER DELGADO, IGNACIO HERNANDO AND MARÍA J. NIETO: Do European primarily Internet banks show scale and experience efficiencies?
- 0413 ÁNGEL ESTRADA, JOSÉ LUIS FERNÁNDEZ, ESTHER MORAL AND ANA V. REGIL: A quarterly macroeconometric model of the Spanish economy.
- 0414 GABRIEL JIMÉNEZ AND JESÚS SAURINA: Collateral, type of lender and relationship banking as determinants of credit risk
- 0415 MIGUEL CASARES: On monetary policy rules for the euro area.
- 0416 MARTA MANRIQUE SIMÓN AND JOSÉ MANUEL MARQUÉS SEVILLANO: An empirical approximation of the natural rate of interest and potential growth. (The Spanish original of this publication has the same number).
- 0417 REGINA KAISER AND AGUSTÍN MARAVALL: Combining filter design with model-based filtering (with an application to business-cycle estimation).
- 0418 JÉRÔME HENRY, PABLO HERNÁNDEZ DE COS AND SANDRO MOMIGLIANO: The short-term impact of government budgets on prices: evidence from macroeconometric models.
- 0419 PILAR BENGOECHEA AND GABRIEL PÉREZ-QUIRÓS: A useful tool to identify recessions in the euro area.
- 0420 GABRIEL JIMÉNEZ, VICENTE SALAS AND JESÚS SAURINA: Determinants of collateral.
- 0421 CARMEN MARTÍNEZ-CARRASCAL AND ANA DEL RÍO: Household borrowing and consumption in Spain:

  A VECM approach
- 0422 LUIS J. ÁLVAREZ AND IGNACIO HERNANDO: Price setting behaviour in Spain: Stylised facts using consumer price micro data.
- 0423 JUAN CARLOS BERGANZA AND ALICIA GARCÍA-HERRERO: What makes balance sheet effects detrimental for the country risk premium?
- 0501 ÓSCAR J. ARCE: The fiscal theory of the price level: a narrow theory for non-fiat money.



